

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Dobrew, Michael; Gerke, Rafael; Giesen, Sebastian; Röttger, Joost

Conference Paper Make-Up Strategies with Incomplete Markets and Bounded Rationality

Beiträge zur Jahrestagung des Vereins für Socialpolitik 2023: Growth and the "sociale Frage"

Provided in Cooperation with: Verein für Socialpolitik / German Economic Association

Suggested Citation: Dobrew, Michael; Gerke, Rafael; Giesen, Sebastian; Röttger, Joost (2023) : Make-Up Strategies with Incomplete Markets and Bounded Rationality, Beiträge zur Jahrestagung des Vereins für Socialpolitik 2023: Growth and the "sociale Frage", ZBW - Leibniz Information Centre for Economics, Kiel, Hamburg

This Version is available at: https://hdl.handle.net/10419/277697

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Make-Up Strategies with Incomplete Markets and Bounded Rationality*

MICHAEL DOBREW[†] RAFAEL GERKE[‡] SEBASTIAN GIESEN[§] JOOST RÖTTGER[¶]

February 15, 2023

Abstract

We study the impact of market incompleteness and bounded rationality on the effectiveness of make-up strategies. To do so, we simulate a heterogeneous-agent New Keynesian (HANK) model with reflective expectations and an occasionally-binding effective lower bound (ELB) on the policy rate. Our simulations show that make-up strategies can mitigate the negative consequences of the ELB for inflation and real economic activity. This result holds both for our HANK model as well as a corresponding representative-agent (RANK) model with complete markets, suggesting that market (in)completeness is not important for the effectiveness of make-up strategies. However, the stabilisation benefits of make-up strategies are small when agents' cognitive ability is consistent with micro-evidence. This result is independent of market (in)completeness, emphasising the importance of rational expectations for make-up strategies. Furthermore, while market incompleteness and bounded rationality complement each other in attenuating the effects of forward guidance in our model, we do not observe such a complementarity with respect to the benefits of make-up strategies.

Keywords: Make-Up Strategies, Incomplete Markets, Bounded Rationality, HANK, Effective Lower Bound, Average Inflation Targeting

JEL Classification: C63, D31, E21, E31, E52, E58, E70

^{*}We thank Klaus Adam, Bence Bardóczy, Keshav Dogra, Refet Gürkaynak, Moritz Lenel, Ralph Lütticke, Ludwig Straub, Harald Uhlig, Maximilian Weiß, Mirko Wiederholt, Andreas Worms, and audiences at the European Central Bank, the Deutsche Bundesbank, the June 2022 Meeting of the Working Group on Econometric Modelling (WGEM), the 2022 Econometric Society European Meeting and the Fall 2022 Midwest Macroeconomics Meeting for helpful comments and suggestions. The views expressed in this paper are those of the authors and do not necessarily represent those of the Deutsche Bundesbank, the European Central Bank or the Eurosystem.

[†]European Central Bank, Directorate General Monetary Policy, Sonnemannstr. 20, 60314 Frankfurt am Main, Germany. Email: michael.dobrew@ecb.europa.eu.

[‡]Deutsche Bundesbank, Directorate General Economics, Wilhelm-Epstein-Str. 14, 60431 Frankfurt am Main, Germany. Email: rafael.gerke@bundesbank.de.

[§]Deutsche Bundesbank, Directorate General Economics, Wilhelm-Epstein-Str. 14, 60431 Frankfurt am Main, Germany. Email: sebastian.giesen@bundesbank.de.

[¶]Deutsche Bundesbank, Directorate General Economics, Wilhelm-Epstein-Str. 14, 60431 Frankfurt am Main, Germany. Email: joost.roettger@bundesbank.de.

1 Introduction

In August 2020, the Federal Reserve announced its revised monetary policy strategy. Among the most striking changes was a shift to a flexible form of average inflation targeting (AIT). In theory, a make-up strategy like AIT can be helpful in a low-rate environment, which is characterised by an elevated risk of nominal rates hitting an effective lower bound (ELB).¹ The key idea behind AIT is that if the inflation target is not met today, monetary policy promises to "make up" for this in the future. If inflation is currently below target, monetary policy commits to steer inflation above target in the future to ensure that the inflation target is met on average. If credible and well understood, this commitment to raise future inflation already lowers expected real rates today and thereby stimulates current economic activity – even when nominal rates are currently stuck at the ELB.

Typically, the benefits of make-up strategies are quite strong in representative-agent New Keynesian (RANK) models, which are frequently used for monetary policy analysis.² However, there are at least two major concerns about the robustness of this finding. First, it is unclear whether the performance of make-up strategies is robust to deviations from rational expectations. Intuitively, if agents do not fully understand that an inflation undershoot today implies an inflation overshoot tomorrow, their behaviour is not going to respond as much as under rational expectations.³ Second, the effectiveness of make-up strategies crucially depends on the willingness and ability of households to substitute consumption over time. By assuming complete markets, RANK models usually exhibit a counterfactually strong sensitivity of household consumption to interest rate changes (see e.g. Vissing-Jørgensen, 2002; Kaplan et al., 2014) and likely exaggerate the power of make-up strategies as a result.

Motivated by these concerns, we study how bounded rationality and market incompleteness influence the effectiveness of make-up strategies. To do so, we develop a quantitative heterogeneous-agent New Keynesian (HANK) model with incomplete markets, heterogeneous households, nominal price and wage rigidities, an occasionally-binding ELB and bounded rationality. The introduction of market incompleteness and borrowing constraints into the New Keynesian model gives rise to household behaviour that is consistent with empirical evidence (see e.g. Kaplan and Violante, 2018). In particular, household consumption becomes less sensitive to interest rate changes and more responsive to changes in temporary income. As a result, under incomplete markets, monetary policy is largely transmitted through indirect general equilibrium effects that affect household income and less via direct effects of interest rate changes (see Kaplan et al., 2018). Market incompleteness may furthermore interact with bounded rationality in a way that can attenuate the macroeco-

¹In principle, unconventional monetary policies, such as asset purchases, forward guidance or negative interest rates, can substitute for further interest rate cuts at an ELB. An alternative to the use of additional instruments is to raise the inflation target, which can reduce the probability of hitting the ELB (see e.g. Coibion et al., 2012; Andrade et al., 2018; Blanco, 2021). We do not discuss these policy options in this paper.

²See e.g. Work Stream on the Price Stability Objective (2021).

³See the literature review at the end of this section for details.

nomic effects of announcements about future interest rates (see Farhi and Werning, 2019). This interaction could thus have immediate consequences for the performance of make-up strategies as well.

We model bounded rationality by assuming that agents form reflective expectations (see García-Schmidt and Woodford, 2019).⁴ The concept of reflective expectations can be viewed as a "smooth" version of level-k thinking, which is supported by experimental evidence (see Farhi and Werning, 2019).⁵ In both cases, agents form beliefs about future aggregate outcomes based on iterative reasoning. More specifically, agents with a certain cognitive ability level believe that the economy behaves as if it were populated by agents that are slightly less intelligent than they are themselves. The higher the cognitive ability level is, the more agents are able to understand the general equilibrium effects of aggregate shocks to the economy. In the limit, as the cognitive ability levels are discrete valued and the belief formation operates in discrete steps (or rounds). By contrast, under reflective expectations, cognitive ability is continuous valued and belief formation follows a differential equation.⁶

To study the impact of bounded rationality and market incompleteness on the performance of make-up strategies, we simulate our model for two monetary policy strategies. The first one is an inflation targeting (IT) strategy, which is modelled via a standard (inertial) Taylor rule commonly used for monetary policy analysis. The second one is an AIT strategy, which we model via a slightly modified Taylor rule that responds to deviations of the average inflation rate from the inflation target. Both interest rate rules are subject to an ELB constraint, which binds endogenously in our model economy that is subject to random aggregate demand and cost-push shocks. To solve and simulate our model, we build on recent advances in the literature that use a sequence-space approximation for heterogeneousagent models (see Boppart et al., 2018; Auclert et al., 2021) and show how to apply this approach to a model with iterative belief formation and an occasionally-binding ELB.

We obtain four main results. First, our model predicts that AIT outperforms IT in terms of inflation and output stabilisation when there is an ELB. By managing agents' expectations, AIT reduces ELB incidences and negative biases in inflation and real economic activity observed under IT. These findings hold for our model regardless of whether markets are complete and agents are rational. Second, although we calibrate our HANK model to match

⁴The model features three types of forward-looking agents: households, intermediate-good firms and labour unions.

 $^{^{5}}$ A RANK model economy with agents that form reflective expectations can be reinterpreted as an economy with a continuum of "level-*k* agents" that are heterogeneous with respect to their cognitive ability (see García-Schmidt and Woodford, 2019). This property will help us identify empirically plausible values for agents' cognitive abilities.

 $^{^{6}}$ A key advantage of using reflective expectations rather than level-*k* thinking as in Farhi and Werning (2019) is that the model is not as susceptible to oscillatory feedback between different rounds of reasoning. Such feedback can cause a model to exhibit explosive behaviour after only a few rounds (see e.g. García-Schmidt and Woodford, 2019; Angeletos and Sastry, 2021; Bianchi-Vimercati et al., 2021). Avoiding such oscillatory behaviour, which can be viewed as an artifact of level-*k* thinking, makes the computation of models more stable, particularly for larger ones.

an empirically plausible average marginal propensity to consume (MPC) out of temporary income, the relative performance of AIT versus IT hardly differs between our HANK model and the corresponding RANK model.⁷ This result suggests that the complete-markets assumption may not be particularly important for the effectiveness of make-up policies. Third, while we find that AIT tends to mitigate the consequences of the ELB better than IT even under reflective expectations, the relative gains become quite small for low cognitive ability levels. Importantly, this includes cognitive ability levels consistent with empirical and experimental evidence. This finding reflects that make-up strategies cannot play out their advantages very well if agents do not fully understand how they operate. Fourth, our model does not indicate that market incompleteness and bounded rationality complement each other in attenuating the ability of make-up strategies to steer the economy. However, our model predicts such a complementarity in the context of interest rate forward guidance, consistent with Farhi and Werning (2019). While these two results may appear to be at odds with each other at first sight, we show how to reconcile them in this paper.

Literature Our paper contributes to three strands of literature. First, it contributes to the growing HANK literature that integrates household heterogeneity and incomplete markets (see İmrohoroğlu, 1992; Huggett, 1993; Aiyagari, 1994) into a New Keynesian environment (see Galí, 2015).⁸ Within the HANK literature, our paper is particularly related to Farhi and Werning (2019), Auclert et al. (2020) and Grimaud (2021), who allow for deviations from rational expectations, as well as to Feiveson et al. (2020), Dobrew et al. (2021) and Djeutem et al. (2022), who look at the interaction between household heterogeneity and make-up policies under rational expectations.⁹ While the last two studies consider an occasionally-binding ELB, the former ones do not. Other papers that study HANK models with an occasionally-binding ELB constraint are Schaab (2020), Fernández-Villaverde et al. (2021), Lee (2021) and McKay and Wieland (2021). However, they restrict attention to standard interest rate rules without make-up elements.

Second, our paper contributes to the literature on make-up monetary policy strategies (see e.g. Svensson, 1999; Arias et al., 2020).¹⁰ Specifically, it contributes to recent studies

⁷As in Hagedorn et al. (2019), we consider a HANK model with nominal wage stickiness and a calibration that attenuates the power of anticipated future monetary accommodation, which AIT relies on. In contrast to these authors, we however allow for positive (nominal) household debt as in Ferrante and Paustian (2019), such that there are redistributive effects that can potentially increase the effectiveness of future interest rate changes by reallocating resources between indebted high- and saving low-MPC households.

⁸See Kaplan and Violante (2018) for a recent survey. Prominent examples of this literature include Werning (2015), McKay et al. (2016), McKay and Reis (2016), Kaplan et al. (2018), Bayer et al. (2019), Hagedorn et al. (2019), Acharya and Dogra (2020), Bhandari et al. (2021), Gornemann et al. (2021) and Luetticke (2021). A complementary literature exists on New Keynesian models with heterogeneous firms and financial frictions (see e.g. Jeenas, 2020; Ottonello and Winberry, 2020; Jungherr et al., 2021).

⁹See also Pfäuti and Seyrich (2022) who introduce cognitive discounting à la Gabaix (2020) into the tractable HANK (THANK) model proposed by Bilbiie (2021). Related, but not within a HANK context, are also Meh et al. (2010) and Bergman et al. (2022). Whereas the former analyse price-level targeting for an overlapping-generations model, the latter study average inflation targeting for a RANK model with heterogeneous workers and a frictional labour market.

¹⁰Price-level targeting as a particular type of make-up strategy is known to have a number of desirable properties in the standard RANK model. For instance, Vestin (2006) shows that, under certain conditions, the

that highlight the limitations of make-up strategies in New Keynesian models with bounded rationality (see e.g. Budianto et al., 2020; Honkapohja and Mitra, 2020; Mele et al., 2020; Erceg et al., 2021; Bodenstein et al., 2022; Dupraz et al., 2022).¹¹ In contrast to these studies, our analysis is not restricted to a RANK environment but allows for household heterogeneity and market incompleteness. Farhi and Werning (2019) show that the interaction between bounded rationality and market incompleteness can strongly attenuate the power of forward guidance, which aims to affect current economic activity by managing agents' expectations. Since this type of expectations management is also at the heart of make-up strategies, an analysis of such strategies that allows for bounded rationality but does not take into account market incompleteness could be misleading. To model bounded rationality, we build on recent work that introduces behavioural or informational frictions into the textbook RANK model to eliminate some of its puzzling features (see Angeletos and Lian, 2018; Woodford, 2018; Farhi and Werning, 2019; García-Schmidt and Woodford, 2019; Gabaix, 2020).¹² As in these papers, we consider forward-looking expectation formation that is limited with respect to how much information agents use. Specifically, we use the approach introduced by García-Schmidt and Woodford (2019) and study its implications for the effectiveness of make-up policies in a HANK context.¹³

Third, from a methodological perspective, our paper contributes to recent studies that solve heterogeneous-agent models by using a local model approximation in the sequence space (see Boppart et al., 2018; Auclert et al., 2021) rather than the recursive state space (see Reiter, 2009; Bayer and Luetticke, 2020). In particular, we show how to leverage the sequence-space approximation to solve a HANK model with non-rational expectations and an occasionally-binding ELB, using anticipated monetary policy shocks (see e.g. Bodenstein et al., 2013; Holden, 2016).¹⁴

Layout The remainder of the paper is organised as follows. Section 2 introduces the model. Section 3 describes the model calibration. Section 4 covers our numerical solution approach. Section 5 presents and discusses the results. Section 6 concludes.

optimal Ramsey policy can be implemented as the optimal discretionary policy if the central bank's objective includes a price-level rather than an inflation target. Furthermore, price-level targeting can approximate the optimal Ramsey policy prescription for ELB episodes (see e.g. Eggertsson and Woodford, 2003).

¹¹See Eusepi and Preston (2018) for a recent survey of monetary policy in RANK models with adaptive learning.

¹²Well-known examples of such features are the forward guidance puzzle (see Del Negro et al., 2015), which states that anticipated future interest rate cuts have implausibly large contemporary effects on economic activity, and the reversal puzzle (see Carlstrom et al., 2015; Gerke et al., 2020), which states that an interest rate peg leads to a counterintuitive contraction in inflation.

¹³Bersson et al. (2019) consider level-*k* thinking as well as reflective expectations for a RANK model with an ELB constraint. In the paper, the authors look at optimal monetary policy and selected experiments, focussing on a deterministic environment. By contrast, we provide a quantitative analysis of a HANK model based on stochastic simulations. Iovino and Sergeyev (2021) study central bank balance sheet policies when agents have reflective expectations.

¹⁴Auclert et al. (2020) also use a sequence-space approximation for a HANK model with non-rational expectations. In contrast to our paper, the model considers sticky household expectations rather than an iterative belief formation and does not consider stochastic simulations with an occasionally-binding ELB constraint.

2 Model

In this section, we extend a standard New Keynesian model with nominal price and wage rigidities (see Galí, 2015) by including idiosyncratic household income risk, incomplete financial markets and non-rational expectations. We also allow for an occasionally-binding effective lower bound (ELB) on the short-term nominal interest rate, which gives rise to ELB episodes of endogenous length and severity. Although we are ultimately interested in analysing the model with random aggregate shocks and non-rational expectations, we provide the model formulation only for deterministic aggregate shocks and do not impose restrictions on how expectations are formed. Building on the model formulation laid out in this section, we study stochastic model simulations under non-rational expectation in Section 5. Since these simulations will be carried out based on a linearised model version (see Section 4), where certainty equivalence holds, we refrain from spelling out how agents form expectations in the presence of aggregate uncertainty for the nonlinear model in a general way.¹⁵

2.1 Economy

This section describes the individual components of the model economy.

Households The economy is populated by a unit-mass continuum of infinitely-lived households, indexed by $i \in [0, 1]$, that derive utility from consumption and leisure. Households face idiosyncratic income risk due to time-varying labour productivity $z_{i,t}$, which affects the size of individual labour earnings, $w_t z_{i,t} N_t$. Idiosyncratic productivity follows a first-order discrete Markov process. We normalise its support to obtain an average productivity level of one for the economy, i.e. $\sum_i \Pr(z_i) z_i = 1$ with $\Pr(z_i)$ denoting the time-invariant share of households with z_i . In each period, households receive firm profits due to ownership of intermediate-good firms in the economy. Shares in these firms are not tradable and thus fully illiquid. Aggregate firm dividends d_t are distributed across households in proportion to their individual productivity relative to average labour productivity.¹⁶ Households can only imperfectly insure themselves against income risk by saving (or borrowing) via a nominal non-state contingent bond, $\tilde{b}_{i,t+s}$, subject to the ad-hoc borrowing constraint $\tilde{b}_{i,t+s} \ge \underline{b}$.¹⁷

Following Farhi and Werning (2019), we assume that households can observe aggregate model variables in the current period t when deciding on their consumption-savings plans. However, for future periods t + s, s > 0, households rely on – potentially non-rational – beliefs for these variables, which is captured by the use of index e. Let Ω_t denote the set

¹⁵See Appendix B for details on how the model formulation in this section relates to the numerical model solution. As in Evans et al. (2022), one could also view the setting presented in this section as one where agents face aggregate uncertainty but rely on point expectations of future aggregate variables.

¹⁶This proportional dividend rule reduces the impact that distributional effects have on aggregate fluctuations (see e.g. Hagedorn et al., 2019).

¹⁷We express individual holdings of nominal bonds in real terms, $\tilde{b}_{i,t} = \tilde{B}_{i,t}/P_t$, where $\tilde{B}_{i,t}$ denotes non-normalised nominal bond holdings and P_t the aggregate price level.

of relevant *endogenous* aggregate variables in period t and Ω_{t+s}^e denote the respective set of period-t beliefs about these variables in period t + s, s > 0. For notational convenience, define the convention that $\Omega_{t+s}^e = \Omega_t$ for s = 0. The tilde symbol used for the sequences $\{\tilde{c}_{i,t+s}\}_{s=0}^{\infty}$ and $\{\tilde{b}_{i,t+s}\}_{s=0}^{\infty}$ indicates that these are plans, made conditional on being in period t and individual states $(b_{i,t-1}, z_{i,t})$, which do not have to materialise in equilibrium, even though consumption and bond holdings are controlled by the household.

Due to sticky nominal wages, household labour supply is entirely demand-determined in the model. As we explain in greater detail below, all household types work the same amount of hours, L_t , which – due to inefficiencies caused by wage dispersion – will not generally coincide with the amount of labour demanded by firms at aggregate real wage w_t , denoted as N_t . Taking as given hours worked and labour demanded, households form beliefs about the respective future values, L_{t+s}^e and N_{t+s}^e , as well as about future values of the aggregate real wage, w_{t+s}^e , aggregate firm dividends, d_{t+s}^e , the (gross) inflation rate, Π_{t+s}^e , and the (gross) nominal interest rate, R_{t+s-1}^e . While households form beliefs about endogenous aggregate variables, they have perfect foresight with respect to the time path of *exogenous* aggregate variables, which are summarised by the vector S_{t+s} that also contains the household discount factor β_{t+s} , $s \ge 0$.

At date *t*, the decision problem of an individual household *i* with bond holdings $b_{i,t-1}$ involves choosing a plan for consumption and bond holdings, $\{\tilde{c}_{i,t+s}, \tilde{b}_{i,t+s}\}_{s=0}^{\infty}$, to maximise expected lifetime utility,

$$\mathbb{E}_{t}\left[\sum_{s=0}^{\infty}\left(\prod_{k=0}^{s}\beta_{t+k-1}\right)\left\{u\left(\tilde{c}_{i,t+s}\right)-v\left(L_{t+s}^{e}\right)\right\}\right],\$$

with felicity functions $u(c) = (c^{1-\sigma} - 1)/(1-\sigma)$ and $v(l) = \chi l^{1+\eta}/(1+\eta)$, subject to the actual budget constraint for the current period t, $\tilde{c}_{i,t} + \tilde{b}_{i,t} = b_{i,t-1}R_{t-1}\Pi_t^{-1} + w_t z_{i,t}N_t + d_t z_{i,t}$, expected future period budget constraints, $\tilde{c}_{i,t+s} + \tilde{b}_{i,t+s} = \tilde{b}_{i,t-1+s}R_{t-1+s}^e(\Pi_{t+s}^e)^{-1} + w_{t+s}^e z_{i,t+s}N_{t+s}^e + d_{t+s}^e z_{i,t+s}$, for s > 0, as well as borrowing limits $\tilde{b}_{i,t+s} \ge \underline{b}$, for $s \ge 0$.¹⁸

It will be useful to define demand functions for individual bond holdings and consumption, given by $x_t(b,z) = x(b_{i,t-1}, z_{i,t}; \{S_{t+s}\}_{s=0}^{\infty}, \Omega_{t-1}, \Omega_t, \{\Omega_{t+1+s}^e\}_{s=0}^{\infty}), x \in \{b, c\}$. These functions solve the household problem at date *t* and constitute the policy functions for $\tilde{b}_{i,t}$ and $\tilde{c}_{i,t}$, respectively. Furthermore, let $\Psi_{i,t-1}$ denote the beginning-of-period joint distribution of bonds $b_{i,t-1}$ and labour productivity $z_{i,t}$ for period *t*. In this paper, there is no public debt and bonds are therefore in zero net supply.¹⁹ The bond market clearing condition hence is given by $\int_i b_{i,t} d\Psi_{i,t-1} = 0$, with $b_{i,t} = b_t(b_{i,t-1}, z_{i,t})$.

Note that without idiosyncratic income risk and borrowing constraints, i.e. if $z_{i,t+s} = 1$,

¹⁸The expectations operator $\mathbb{E}_t [\cdot]$ computes household expectations with respect to idiosyncratic labour productivity.

¹⁹In RANK models, lump-sum taxes ensure Ricardian equivalence holds, allowing a separation of monetary and fiscal policy in the presence of positive public debt. Such a separation is no longer possible in HANK models. Even if monetary policy transmission does not depend on fiscal policy in the corresponding RANK model, fiscal policy and the details of how it is specified would play a non-trivial role in our HANK model (see e.g. Kaplan et al., 2018). Given that this would make it more complicated to understand the role of market (in)completeness for make-up strategies, we disregard fiscal policy in this paper altogether.

for $s \ge 0$, as well as $\underline{b} = -\infty$, individual consumption $\tilde{c}_{i,t}$ can be expressed in closed form as

$$\tilde{c}_{i,t} = \frac{b_{i,t-1} \frac{R_{t-1}}{\Pi_t} + w_t N_t + d_t + \sum_{s=1}^{\infty} \left(\prod_{u=1}^s \left(\frac{R_{t+u-1}^e}{\Pi_{t+u}^e} \right)^{-1} \right) \left(w_{t+s}^e N_{t+s}^e + d_{t+s}^e \right)}{1 + \sum_{s=1}^{\infty} \prod_{u=1}^s \beta_{t+u-1}^{1/\sigma} \left(\frac{R_{t+u-1}^e}{\Pi_{t+u}^e} \right)^{1/\sigma-1}},$$

which illustrates that individual consumption in period *t* is a fraction of the sum of initial wealth and the present-value of lifetime income as perceived by the household at date *t*.²⁰ Given $\tilde{c}_{i,t}$, bond holdings $\tilde{b}_{i,t}$ are then determined residually via the household period budget constraint $\tilde{b}_{i,t} = b_{i,t-1}R_{t-1}\Pi_t^{-1} + w_t z_{i,t}N_t + d_t z_{i,t} - \tilde{c}_{i,t}$.

Goods market A representative firm combines intermediate production inputs $Y_{j,t}$, supplied by a unit-mass continuum of monopolistic firms, indexed by $j \in [0,1]$, to produce the final good $Y_t = (\int_j Y_{j,t}^{(\theta_p - 1)/\theta_p} dj)^{\theta_p/(\theta_p - 1)}$. Profit maximisation by the firms yields $Y_{j,t} = (P_{j,t}/P_t)^{-\theta_p}Y_t$ as the demand function for intermediate good j. The price of the final good is $P_t = (\int_j P_{j,t}^{1-\theta_p} dj)^{1/(1-\theta_p)}$, where $P_{j,t}$ denotes the price of input $Y_{j,t}$ set by intermediate-good firm j. The intermediate good $Y_{j,t}$ is produced by firm j with production function $Y_{j,t} = N_{j,t}$. Real marginal costs are equal across firms and given by w_t .

Intermediate-good firms set prices subject to a Calvo-type friction. Only a randomly selected fraction $(1 - \xi_p)$ of firms is therefore able to reset its price $P_{j,t}$ in each period, taking into account future price stickiness. For the remaining fraction of firms that cannot optimally reset their prices (ξ_p) , the intermediate-good price is updated according to $P_{j,t} = P_{j,t-1} \left[\Pi^{l_p} (\Pi_{t-1}^e)^{1-l_p} \right]$, where Π denotes the long-run inflation rate and $t_p \in [0,1]$ how prices are indexed. When setting its price at date *t*, firm *j* can observe the endogenous variables for the current period but relies on beliefs about their future values. Specifically, the relevant beliefs involve the aggregate price level, P_{t+s}^e , aggregate production, Y_{t+s}^e , and the aggregate real wage, w_{t+s}^e . Firms do, however, possess perfect foresight with respect to fluctuations in exogenous aggregate variables. For simplicity, we assume that firms discount future profits with constant discount factor β , which is the long-run value of household discount factor β_t . We assume that intermediate firms have to pay a fixed cost $\Phi \ge 0$ in each period.

An intermediate-good firm j that can reset its price in period t chooses $P_{j,t}$ to maximise

$$\sum_{s=0}^{\infty} \left(\beta \xi_p\right)^s \left[\left(P_{j,t+s}/P_{t+s}^e - \lambda_{t+s} w_{t+s}^e\right) Y_{j,t+s} - \Phi \right],$$

subject to the demand function $Y_{j,t} = (P_{j,t}/P_t)^{-\theta_p}Y_t$ and the indexing rule $P_{j,t+s} = P_{j,t} \times \prod_{u=1}^{s} \left[\prod_{t=u-1}^{l_p} (\prod_{t=u-1}^{e})^{1-\iota_p} \right]$, which updates prices until they can again be re-set in the future. We introduce the random variable λ_t to allow for cost-push shocks.²¹

We focus on a symmetric equilibrium in which all firms that can reset their prices do so in the same way. Let P_t^* denote the common reset price and define $\Pi_t^* = P_t^*/P_{t-1}$ as well

 $^{^{20}}$ The derivation of the optimal decision rules shown in this section can be found in Appendix A.

²¹As in Chen et al. (2012), we assume that the mark-up shifter λ_t shows up in the intermediate-good firm problem, affecting price setting, but not in actual profits. This formulation is similar to how Hagedorn et al. (2019) treat Rotemberg-type price adjustment costs in the goods market.

as $\Pi_t = P_t/P_{t-1}$. It is straightforward to show that the first-order condition for P_t^* can be written as

$$\Pi_{t}^{*} = \frac{\theta_{p}}{\theta_{p} - 1} \frac{\lambda_{t} w_{t} Y_{t} + \sum_{s=1}^{\infty} (\beta \xi_{p})^{s} \left(\prod_{u=1}^{s} \frac{\Pi^{i_{p}} (\Pi_{t+u-1}^{e})^{1-i_{p}}}{\Pi_{t+u}^{e}} \right)^{-\theta_{p}} \lambda_{t+s} w_{t+s}^{e} Y_{t+s}^{e}}{Y_{t+s}} \Pi_{t},$$

with inflation Π_t satisfying $\Pi_t^{1-\theta_p} = (1-\xi_p)(\Pi_t^*)^{1-\theta_p} + \xi_p (\Pi_t^{\iota_p}(\Pi_{t-1})^{1-\iota_p})^{1-\theta_p}$.

Due to price dispersion, $\mu_{p,t}$, aggregate output in the economy satisfies $\mu_{p,t}Y_t = N_t$ and is hence distorted for $\mu_{p,t} > 1$. The law of motion for price dispersion is given by $\mu_{p,t} = \prod_t^{\theta_p} \left[(1 - \xi_p)(\Pi_t^*)^{-\theta_p} + \xi_p \left(\prod_{t=1}^{\iota_p} (\prod_{t=1}^{e})^{1-\iota_p} \right)^{-\theta_p} \mu_{p,t-1} \right]$. Goods market clearing requires $Y_t = \int_i c_{i,t} d\Psi_{i,t-1} + \Phi$, where $c_{i,t} = c_t (b_{i,t-1}, z_{i,t})$. Finally, aggregate firm profits are $d_t = Y_t - w_t N_t - \Phi$.

Labour market There is a unit-one continuum of labour unions, indexed by $k \in [0, 1]$, that aggregate labour supplied by households to specialised labour services $N_{k,t}$ which are then offered to a representative labour packer at nominal wage $W_{k,t}$. These labour services are in turn combined by the labour packer to generate the composite labour input N_t according to a CES function, $N_t = (\int_k N_{k,t}^{(\theta_w - 1)/\theta_w} dk)^{\theta_w/(\theta_w - 1)}$, and sold to intermediate-good firms at the nominal wage W_t .²² Profit maximisation by the labour packer yields the demand function for labour services of union k, $N_{k,t} = (W_{k,t}/W_t)^{-\theta_w}N_t$. The aggregate wage satisfies $W_t = (\int_k W_{k,t}^{1-\theta_w} dk)^{1/(1-\theta_w)}$.

Each union k generates labour services by aggregating efficiency units of labour supplied by households, $N_{k,t} = \int_i z_{i,t} n_{i,k,t} d\Psi_{i,t-1}$. Household *i* is employed by union k according to the uniform rule $n_{i,k,t} = N_{k,t}$, implying that all households supply the *same* amount of labour to union k and hence in total. Total hours worked by household *i* are $l_{i,t} = \int_k n_{i,k,t} dk$ and – as will be shown below – equal across households, i.e. $l_{i,t} = L_t$.

Unions are subject to nominal rigidities à la Calvo when setting wage $W_{k,t}$ for labour services $N_{k,t}$. Similar to the goods market, only a fraction $1 - \xi_w$ of randomly selected unions can thus optimally reset the wage $W_{k,t}$ in each period, whereas the remaining fraction cannot re-optimise and updates the wage based on the indexing rule $W_{k,t+s} = W_{k,t} \times$ $\prod_{u=1}^{s} \left[\Pi^{t_w} (\Pi_{t+u-1}^e)^{1-t_w} \right]$, with $t_w \in [0,1]$. Like the households and the intermediate-good firms, labour unions form beliefs about endogenous aggregate variables, $\{\Omega_{t+1+s}^e\}_{s=0}^{\infty}$, but have perfect foresight regarding exogenous aggregate variables, $\{S_{t+s}\}_{s=0}^{\infty}$.

Taking as given household consumption-saving decisions and anticipating future wage stickiness, a union k that can reset its wage at date t chooses $W_{k,t}$ to maximise

$$\sum_{s=0}^{\infty} (\beta \xi_w)^s \left\{ M U_{t+s}^e (W_{k,t+s}/P_{t+s}^e) N_{k,t+s} - M V_{t+s}^e N_{k,t+s} \right\},\$$

with $MU_{t+s} = \int z_{i,t+s} c_{i,t+s}^{-\sigma} d\Psi_{i,t+s-1}$ and $MV_{t+s} = \int \chi l_{i,t+s}^{\eta} d\Psi_{i,t+s-1}$, subject to labour ser-

²²The structure of the labour market follows Auclert et al. (2020), who extend the sticky-wage framework developed in Erceg et al. (2000) and Schmitt-Grohé and Uribe (2006) to a HANK model.

vice demand $N_{k,t+s} = (W_{k,t+s}/W_{t+s}^e)^{-\theta_w} N_{t+s}^e$ and the indexing rule for wages.

In a symmetric equilibrium, all unions that can reset their wages choose to do so in the same way. Define W_t^* as the common nominal reset wage for these unions. The unions' first-order condition for the nominal wage W_t^* can be written in real terms as

$$w_{t}^{*} = \frac{\theta_{w}}{\theta_{w} - 1} \frac{N_{t} w_{t}^{\theta_{w}} M V_{t} + \sum_{s=1}^{\infty} (\beta \xi_{w})^{s} \left(\prod_{u=1}^{s} \frac{\Pi^{t_{w}} (\Pi^{e}_{t+u-1})^{1-t_{w}}}{\Pi^{e}_{t+u}}\right)^{-\theta_{w}} N_{t+s}^{e} (w_{t+s}^{e})^{\theta_{w}} M V_{t+s}^{e}}{N_{t} w_{t}^{\theta_{w}} M U_{t} + \sum_{s=1}^{\infty} (\beta \xi_{w})^{s} \left(\prod_{u=1}^{s} \frac{\Pi^{t_{w}} (\Pi^{e}_{t+u-1})^{1-t_{w}}}{\Pi^{e}_{t+u}}\right)^{1-\theta_{w}} N_{t+s}^{e} (w_{t+s}^{e})^{\theta_{w}} M U_{t+s}^{e}}$$

where $w_t^* = W_t^* / P_t$.

Using the uniform labour demand rule $n_{i,k,t} = N_{k,t}$ and the demand function for labour services offered by union k, $N_{k,t} = (W_{k,t}/W_t)^{-\theta_w}N_t$, total hours for household i, $l_{i,t} = \int_k n_{i,k,t}dk$, can be written as $l_{i,t} = \mu_{w,t}N_t$, with wage dispersion $\mu_{w,t}$ evolving according to the law of motion $\mu_{w,t} = (1 - \xi_w) (w_t^*/w_t)^{-\theta_w} + \xi_w (w_{t-1}/w_t)^{-\theta_w} (\Pi^{\iota_w} (\Pi_{t-1}^e)^{1-\iota_w}/\Pi_t)^{-\theta_w} \mu_{w,t-1}$. It is straightforward to see that – as noted earlier – all households work the same amount of hours $l_{i,t} = L_t$, which is different from composite labour input N_t for $\mu_{w,t} > 1$. It therefore holds that $L_t = \mu_{w,t}N_t$. Finally, the aggregate real wage satisfies $w_t^{1-\theta_w} = (1 - \xi_w) (w_t^*)^{1-\theta_w} + \xi_w (\Pi^{\iota_w} (\Pi_{t-1}^e)^{1-\iota_w} w_{t-1}/\Pi_t)^{1-\theta_w}$.

Monetary policy The central bank sets the nominal interest rate R_t based on an instrument rule and subject to an ELB, $R_t = \max{\{\tilde{R}_t, R_{ELB}\}}$. For the shadow rate \tilde{R}_t , we consider a slightly modified inertial Taylor rule,

$$\frac{\tilde{R}_t}{R} = \left(\frac{\tilde{R}_{t-1}}{R}\right)^{\rho_R} \left(\left(\frac{\Pi_t^{(T)}}{\Pi}\right)^{\phi_{\Pi}} \left(\frac{Y_t}{Y}\right)^{\phi_Y} \right)^{1-\rho_R},$$

where *R* denotes the (quarterly) long-run nominal interest rate, Π the (quarterly) inflation target, *Y* the long-run output level, and $\Pi_t^{(T)} = (\prod_{k=1}^T \Pi_{t-k+1})^{1/T}$ the average (quarterly) inflation rate over the past *T* periods.

This rule captures two types of monetary policy strategies. For T = 1, the rule reduces to a standard Taylor rule that targets the current inflation rate. We refer to this case as an inflation targeting (IT) strategy. This is the case usually considered in the New Keynesian literature to model monetary policy. For T > 1, the policy rule targets the average inflation rate over the last T periods. We refer to this case as an average inflation targeting (AIT) strategy. Under this specification, missing the inflation target in a given period has implications for how interest rates will be set in future periods. Specifically, monetary policy is committed to make up for past inflation misses until the average inflation rate meets the target again. As is well known (see Svensson, 1999; Arias et al., 2020), this make-up element can help to stabilise inflation more effectively by guiding agents' expectations, particularly when monetary policy is constrained by an ELB.

As a measure of real economic activity, output enters the interest rate rule as well. This assumption is standard and implies that the central bank does not follow a strict (A)IT strat-

egy. Instead, it follows a flexible version of it that reflects potential policy trade-offs, e.g. due to a dual mandate. This is consistent with the monetary policy of the Fed, both before and after its recent strategy revision. The interest rate rule also includes an interest rate smoothing component, which captures that interest rates tend to be adjusted only gradually in practice (see e.g. Nakata and Schmidt, 2019).

2.2 Equilibrium

In this section, we define the model equilibrium. Having derived the relevant equilibrium conditions for arbitrary beliefs in the previous section, we start by defining a temporary equilibrium for the model.²³ Building on this, we then define the equilibrium under rational and reflective expectations.

Temporary equilibrium We define a temporary equilibrium as follows. Let $\Omega = (\mu_p, \mu_w, \Pi, \{\Pi_-\}, \Pi^*, \Psi(\cdot), d, L, MU, MV, N, R, \tilde{R}, w, w^*, Y)$ denote the set of endogenous aggregate variables and $S = (\beta, \lambda)$ the set of exogenous aggregate variables.²⁴ Given initial values Ω_{t-1} , S_{t-1} and beliefs $\{\Omega_{t+1+s}^e\}_{s=0}^{\infty}$, a temporary equilibrium consists of sequences $\{S_{t+s}, \Omega_{t+s}, b_{t+s}(\cdot), c_{t+s}(\cdot)\}_{s=0}^{\infty}$, such that households, firms and unions solve their respective optimisation problems, markets clear and the wealth distribution is consistent with household actions at every date $t + s, s \ge 0$.

Rational-expectations equilibrium The rational-expectations equilibrium provides us with an important benchmark for our model analysis under non-rational expectations. We can define a rational-expectations equilibrium simply as a temporary equilibrium that satisfies perfect foresight (see Farhi and Werning, 2019). Specifically, a rational-expectations equilibrium is given by $\{S_{t+s}, \Omega_{t+s}, b_{t+s}(b, z), c_{t+s}(b, z), \Omega_{t+s}^{e}\}_{s=0}^{\infty}$, such that $\{S_{t+s}, \Omega_{t+s}, b_{t+s}(b, z), c_{t+s}(b, z), \Omega_{t+s}^{e}\}_{s=0}^{\infty}$ and $\{\Omega_{t+s}^{e}\}_{s=0}^{\infty} = \{\Omega_{t+s}\}_{s=0}^{\infty}$ holds for $s \ge 0$ (perfect foresight).

Reflective-expectations equilibrium To model bounded rationality, we consider a reflective-expectations equilibrium (see García-Schmidt and Woodford, 2019). The concept of reflective expectations is closely related to level-k thinking, which is analysed in a HANK context by Farhi and Werning (2019). In both cases, beliefs are updated based on a recursive scheme after an aggregate shock is realised. While both concepts are closely related, level-k thinking is more well known and easier to understand than the more general concept of

²³The temporary equilibrium concept goes back to Hicks (1939), Lindahl (1939) and Grandmont (1977). More recently, Preston (2005) and Woodford (2013), as well as Farhi and Werning (2019) and García-Schmidt and Woodford (2019), have applied this concept for the analysis of New Keynesian models with bounded rationality.

²⁴By including the lagged inflation rates $\{\Pi_{-s,t}\}_{s \in \{1,2,\dots,T-2\}}$, where $\Pi_{-s,t} = \Pi_{-(s-1),t-1}$ for $s \in \{1,2,\dots,T-2\}$ and $\Pi_{-0,t-1} = \Pi_{t-1}$, in the vector of endogenous variables, Ω_{t-1} and S_t contain all state variables needed to determine Ω_t , since we can write $\Pi_t^{(T)} = (\Pi_t \times \prod_{s=0}^{T-2} \Pi_{-s,t-1})^{1/T}$.

reflective expectations. We thus first describe level-k thinking in the context of the model and then turn to reflective expectations.

Under level-*k* thinking, the beliefs of agents with the discrete level of reasoning k > 1 about outcomes in period t + s, s > 0, denoted as $\Omega_{t+s}^{e,k}$, are given by the equilibrium outcomes for that period in an economy where all agents possess the level of reasoning k - 1, denoted as Ω_{t+s}^{k-1} .²⁵ Starting with some initial beliefs held by agents in a "level-1 economy" in period *t* about outcomes in period t + s, $\Omega_{t+s}^{e,1}$, beliefs held by "level-*k* agents" in period *t* about outcomes in period t + s, $\Omega_{t+s}^{e,1}$, beliefs held by "level-*k* agents" in period *t* about outcomes in period t agent responds to changes in the exogenous aggregate variables (e.g. an increase in patience) but fails to take into account any general equilibrium effects of the shock. For k > 1, level-*k* agents take general equilibrium effects into account but only imperfectly. The more rounds of reasoning *k* an agent goes through, the closer his beliefs approaching rational expectations for $k \to \infty$.

Reflective expectation formation operates similarly but does not proceed in discrete steps. Instead, beliefs are updated continuously. Let *n* denote the – now continuous – level of cognitive ability under reflective expectations. Beliefs about outcomes in period t + s held by agents with cognitive ability level *n* in period *t*, denoted as $\Omega_{t+s}^{e,n}$, are now formed based on the differential equation $d\Omega_{t+s}^{e,n}/dn = \Omega_{t+s}^n - \Omega_{t+s}^{e,n}$, where Ω_{t+s}^n denotes equilibrium outcomes in period t + s for an economy populated by *n*-type agents.

To illustrate the connection between reflective expectations and level-k thinking, it is helpful to consider a discrete approximation of the continuous updating process,

$$\Omega_{t+s}^{e,n+dn} \cong dn \times \Omega_{t+s}^n + (1-dn) \times \Omega_{t+s}^{e,n}$$

where dn > 0 is a very small real number. Beliefs of agents with cognitive ability n + dnabout outcomes in period t + s can hence be viewed as a convex combination of equilibrium outcomes in period t + s for an economy populated by *n*-type agents, Ω_{t+s}^n , and beliefs for period t + s held by agents in such an economy, $\Omega_{t+s}^{e,n}$. Compared to level-*k* thinking, which is effectively nested for dn = 1 in the expression above, updating of reflective beliefs occurs in a smooth manner since $dn \approx 0$ implies an updating weight of approximately one for $\Omega_{t+s}^{e,n}$. This smoothness avoids jumps in expectations that can take place during belief updating under level-*k* thinking and result in oscillatory behaviour of the economy over time (see García-Schmidt and Woodford, 2019). As noted by Angeletos and Sastry (2021), such behaviour can be considered an artifact of discrete belief updating under level-*k* thinking without deeper economic meaning.²⁶

Formally, we can define the reflective-expectations equilibrium as follows. Let $\{S, \Omega, b(b,z), c(b,z), \Omega^e\}_{t=0}^{\infty}$ be a rational-expectations equilibrium where all variables stay at the steady state forever. Define initial beliefs $\{\Omega_{t+s}^{e,0}\}_{s=0}^{\infty} = \{\Omega\}_{s=0}^{\infty}$, such that, in period

²⁵Equilibrium outcomes in period t + s in turn depend on beliefs for all periods going forward, i.e. $\{\Omega_{t+s+j}^{e,k-1}\}_{j=0}^{\infty}$.

²⁶In addition to eliminating this model feature, the smooth belief formation under reflective expectations also improves the stability of the numerical model solution.

t, the economy is expected to remain at the steady state in all future periods.²⁷ Given a sequence $\{S_{t+s}\}_{s=0}^{\infty}$, a reflective-expectations equilibrium consists of a belief-updating process, $d\Omega_{t+s}^{e,n}/dn = \Omega_{t+s}^n - \Omega_{t+s}^{e,n}$, and sequences indexed by level of reflective reasoning $n \ge 0$, $\{S_{t+s}, \Omega_{t+s}^n, b_{t+s}^n(\cdot), c_{t+s}^n(\cdot)\}_{s=0}^{\infty}$, such that each $\{S_{t+s}, \Omega_{t+s}^n, b_{t+s}^n(\cdot), c_{t+s}^n(\cdot)\}_{s=0}^{\infty}$ constitutes a temporary equilibrium for beliefs $\{\Omega_{t+s}^{e,n}\}_{s=0}^{\infty}$, with $n = n^* > 0$ being the actual level of reflective reasoning in the economy. For notational convenience, we use the convention $\Omega_{t+s} = \Omega_{t+s}^{n^*}$, $s \in \{0, 1, 2, ..., T\}$, for the reflective-expectations case. Rational expectations are nested as a special case for the limit $n^* \to \infty$.

García-Schmidt and Woodford (2019) show that period-*t* beliefs about outcomes in period t + s held by agents in a reflective-expectations equilibrium with the continuous level of reasoning n, $\Omega_{t+s}^{e,n}$, are equal to the average beliefs held in a level-*k* economy where agents with the discrete level of reasoning $k \ge 1$ are distributed across the population according to a Poisson distribution with mean n,

$$\Omega_{t+s}^{e,n} = \sum_{k=1}^{\infty} \frac{n^{k-1} \exp\left(-n\right)}{(k-1)!} \Omega_{t+s}^{e,k}.$$

The beliefs held by level-*k* thinkers for period t + s, $\Omega_{t+s}^{e,k}$, are in this case determined by the discrete recursion $\Omega_{t+s}^{e,k} = \Omega_{t+s}^{k-1}$.

3 Calibration

We calibrate the model for the United States at a quarterly frequency. The parameter values are summarised in Table 1. Parameters related to the monetary policy rule are specified as follows. We assume an annual inflation target (Π)⁴ of 2%, an effective lower bound (R_{ELB})⁴ of 0.125% (see Nakata and Schmidt, 2019) and the rather standard policy rule parameters ($\phi_{\Pi}, \phi_Y, \rho_R$) = (1.5, 1, 0.75).²⁸ Consistent with estimates reported in Bernanke (2020), we target an annual long-run real rate of 1% and implement it via the (long-run) household discount factor. For the RANK model, the required β -value is 0.998, whereas it is 0.993 for the HANK model.²⁹

We choose a standard value of 2 for the coefficient of relative risk aversion σ and a Frisch elasticity $1/\eta$ of 0.5 (see Chetty et al., 2011). For the RANK model, we normalise the labour disutility parameter χ to 1 and set it equal to 1.023 for the corresponding HANK model, such that both model versions display the same steady-state output level. The house-hold borrowing limit is set to -1.667, which equals 5 times the monthly household labour income (see Ferrante and Paustian, 2019). We set the elasticity of substitution between in-

²⁷It is of course possible to consider initial beliefs based on more general rational-expectations equilibria (see e.g. Farhi and Werning, 2019). We assume this particular equilibrium because our numerical model solution makes this assumption as well.

²⁸As suggested by Brayton et al. (1997) and Taylor (1999), we use an output response coefficient of 1, which is somewhat higher compared to parameter values often considered in the literature (see e.g. Taylor, 1993). See also Arias et al. (2020).

²⁹The discount factor is lower for the HANK model calibration because the households' precautionary saving motive pushes down the real rate for a given discount factor (see e.g. Huggett, 1993).

Parameter	Description	Value
β	Discount factor	0.993*
χ	Weight labour disutility	1.023*
Φ	Fixed cost of production	0.167
ϕ_{Π}	Inflation coeff. policy rule	1.5
ϕ_Y	Output coeff. policy rule	1
η	Inverse Frisch elasticity	2
ι_p	Price indexation	1
ι_w	Wage indexation	1
λ	Price mark-up shifter	1
$(\Pi)^4$	Long-run inflation target (annual)	1.02
Π_{w}	Long-run nominal wage inflation	1.005
$oldsymbol{ heta}_p$	Price elast. of subst. int. goods	6
$\hat{\theta_w}$	Wage elast. of subst. labour services	6
$ ho_eta$	Persist. discount factor	0.85
ρ_{λ}	Persist. mark-up shifter	0
$ ho_z$	Persist. idiosync. productivity	0.966
$ ho_R$	Interest rate smoothing	0.75
σ	Coeff. of relative risk aversion	2
$\sigma_{\!eta}$	Std. dev. discount factor shock	0.006
σ_{λ}	Std. dev. mark-up shock	0.118
σ_{z}	Std. dev. idiosync. prod. shock	0.052
ξ_p	Calvo price-setting	0.85
ξw	Calvo wage-setting	0.85
\underline{b}	Household borrowing limit	-1.667
$(R_{ELB})^4$	ELB (annual)	1.001

 Table 1: Model parameters for HANK model

* For the RANK model, we use $(\beta, \chi) = (0.998, 1)$ instead.

termediate goods θ_p and the Calvo price-setting parameter ξ_p to 6 and 0.85, respectively (see McKay et al., 2016), and normalise the degree of price indexation to $\iota_p = 1$. For the labour market, we use the same parameter values as for the goods market, i.e. $\theta_w = \theta_p$, $\xi_w = \xi_p$ and $\iota_w = \iota_p$. As in Hagedorn et al. (2019), we calibrate the fixed production cost to obtain zero firm profits in the steady state, which is accomplished with $\Phi = 0.167$.

Idiosyncratic productivity $z_{i,t}$ follows a log-normal AR(1) process with persistence parameter ρ_z and shock standard deviation σ_z . We set $\rho_z = 0.966$ (see McKay et al., 2016) and – similar to Fernández-Villaverde et al. (2021) – calibrate the shock standard deviation to match an average annual marginal propensity to consume (MPC) of 0.55 (see Auclert et al., 2020), implying $\sigma_z = 0.052$.³⁰ We also consider log-normal AR(1) processes for the household discount factor and the mark-up shifter, given by $\beta_t = \beta^{1-\rho_\beta} \beta_{t-1}^{\rho_\beta} \exp(\varepsilon_{\beta,t})$ and $\lambda_t = \lambda^{1-\rho_\lambda} \lambda_{t-1}^{\rho_\lambda} \exp(\varepsilon_{\lambda,t})$, respectively. The i.i.d. shock $\varepsilon_{x,t}$ is drawn from a normal distribution with zero mean and standard deviation σ_x , $x \in \{\beta, \lambda\}$. The persistence parameters

³⁰Our model calibration yields a Gini coefficient of 0.80 in the steady state, which is slightly lower than the value of 0.87 reported by Kuhn and Ríos-Rull (2016) based on 2013 US data on liquid wealth.



Figure 1: Distribution of level-k agents for different cognitive ability levels n^*

 ρ_{β} and ρ_{λ} are set to 0.85 and 0, respectively (see Nakata and Schmidt, 2019). We calibrate the shock standard deviations to match statistics for the rational-expectations RANK model with an IT interest rate rule, i.e. for T = 1. To obtain an ELB frequency of 20%, we set σ_{β} to 0.0062. For σ_{λ} , we choose a value of 0.118 to match the standard deviation of annualised US inflation for the time period 1997:Q3-2017:Q2, which is 0.52% (see Nakata and Schmidt, 2019).³¹ When analysing AIT, we consider specifications with averaging windows of T = 16 and T = 32 quarters, i.e. 4 and 8 years, for the average inflation target.³² We refer to these specifications as AIT4 and AIT8, respectively.

For the reflective-expectations case, we consider the cognitive ability levels $n^* \in \{0.5, 1, 2, 3, 6\}$. Based on the interpretation proposed by García-Schmidt and Woodford (2019) (see Section 2.2), different n^* -values imply different (Poisson) distributions of level-*k* agents in the population. Figure 1 displays the population shares associated with the values that we consider for the simulation exercise. Experimental evidence typically suggests that most probability mass is covered by $k \in \{1, 2, 3\}$ (see e.g. Nagel, 1995; Costa-Gomes and Crawford, 2006; Arad and Rubinstein, 2012), which is consistent with the distributions predicted by $n^* \in \{0.5, 1, 2\}$.³³ As argued by Farhi and Werning (2019), experimental evidence relies

³¹For simplicity, both shock standard deviations are calibrated to match the respective targets under the assumption that the other aggregate shock is not operative. They are hence not jointly calibrated.

³²The remaining model parameters are kept unchanged in this case.

³³There are, however, exceptions. For example, Kneeland (2015)'s experimental evidence shows substantial

on games that are substantially simpler than the environment agents are facing in macroeconomic models. As a result, the *k*-values relevant for our quantitative model are likely lower than the ones found in the experimental literature. Whereas the n^* -values 0.5, 1 and – perhaps – 2 can be considered empirically plausible, the values 3 and 6 suggest a level of cognitive ability that is hard to reconcile with (micro-level) evidence. Nevertheless, we consider these higher values as well to investigate how large n^* has to be in order for the economy to behave as in the rational-expectations case.

4 Numerical solution

In theory, the extended path method is a natural candidate for stochastic simulations of a model with reflective expectations.³⁴ However, the method is not well suited for our particular application, which not only involves iterative belief formation but also an ELB and household heterogeneity.³⁵ In a stochastic simulation, the model economy is hit with new unanticipated aggregate shocks in each simulation period going forward. When using the extended path method for such a simulation, one has to numerically compute a new nonlinear transition path back to the steady state for each simulation period. In practice, these computations become already quite slow for our rather simple RANK model when allowing for iterative belief formation and an occasionally-binding ELB. In a model that also includes incomplete markets, such that the entire wealth distribution becomes a relevant aggregate state variable, these speed issues become prohibitive.

To overcome this difficulty, we use a numerical solution approach that is based on a local approximation of the model in the sequence space (see Boppart et al., 2018; Auclert et al., 2021) and takes into account the occasionally-binding ELB constraint by using anticipated monetary policy shocks (see Bodenstein et al., 2013; Holden, 2016).³⁶ The algorithm consists of two steps. The first step involves the computation of impulse response functions for the endogenous aggregate variables to shocks to the exogenous aggregate model variables. These shocks hit the economy in the deterministic steady state and are assumed to be unanticipated, transitory and small. Under these assumptions, Boppart et al. (2018)

³⁵We cannot use standard perturbation techniques (see e.g. Schmitt-Grohé and Uribe, 2004; Reiter, 2009) for the HANK model with reflective expectations due to the iterative nature of belief formation.

probability mass also at k = 4, which is in line with the distribution implied by $n^* = 2$. Similarly, based on firm survey data, Coibion et al. (2021) find levels of reasoning $k \in \{3,4,5\}$ to be well represented.

 $^{^{34}}$ See Heer and Maußner (2009) for a recent treatment of the method. As discussed in García-Schmidt and Woodford (2015), there is a direct link between the belief formation process under reflective expectation (or level-*k* thinking) and the extended path method, which was pioneered by Fair and Taylor (1983). However, García-Schmidt and Woodford (2015, 2019), as well as Farhi and Werning (2019) and Bianchi-Vimercati et al. (2021), only consider one-time MIT shocks and do not employ the extended path algorithm to perform model simulations with randomly drawn aggregate shocks. Assuming that agents rely on point expectations of future aggregate variables, which effectively implies certainty equivalence, it is straightforward to set up the extended path algorithm based on the equilibrium definitions from Section 2 (see Appendix B). See Qiu (2018) for an alternative approach that uses a recursive model formulation for a New Keynesian model with level-*k* thinking and reduced-form household heterogeneity.

³⁶McKay and Wieland (2021) also use anticipated monetary policy shocks to enforce the ELB for simulations of a continuous-time HANK model with durable consumption.

	ELB incidence		Infla	Inflation (%)		put (%)
Policy rule	Freq.	Avg. duration	Ανσ	Std	Ανσ	Std
Toney fule	(%)	(quarters)	1105.	Std.	1115.	Bid.
IT (w/o ELB)	0	0	2.00	0.78	0.00	0.88
AIT4 (w/o ELB)	0	0	2.00	0.70	0.00	0.95
AIT8 (w/o ELB)	0	0	2.00	0.69	0.00	0.97
IT	20.18	6.78	1.95	0.87	-0.11	1.17
AIT4	17.11	6.65	1.98	0.74	-0.07	1.14
AIT8	16.71	6.42	1.99	0.72	-0.06	1.14

Table 2: Results for RANK model with rational expectations (RE-RANK)

show that the computed impulse response functions can be viewed as a linear approximation of the model in the sequence space. In the second step, we use this remarkable result to perform Monte Carlo simulations for our model based on the linearised model. Given that the model approximation is a local one, it does not take into account the ELB constraint on the nominal interest rate. To enforce the ELB for the model simulations, we use news shocks to the policy rate (see Holden, 2016). In principle, applying this method requires the assumption of perfect foresight to be satisfied. However, we show how to apply it to our model with reflective expectations and demonstrate the numerical accuracy of the approach. Details about our numerical solution and its accuracy can be found in Appendix **B**.

5 Results

This section presents and discusses the results of our quantitative model analysis.

5.1 Stochastic simulations

In this section, we explore the impact of market incompleteness and bounded rationality on the effectiveness of AIT vis-à-vis IT based on stochastic simulations.³⁷ To isolate the contribution of the two features, we proceed in four steps. To establish a baseline, we start with a comparison of IT and AIT for the standard RANK model with complete markets and rational expectations (RE-RANK). Then we relax the complete-markets assumption and compare the two strategies for the rational-expectations HANK (RE-HANK) model. After that, we provide a comparison of IT and AIT for the RANK model with bounded rationality (BR-RANK), highlighting the role played by the rational-expectations assumption. Finally, we compare the two monetary policy strategies for the full (BR-HANK) model, abandoning both complete markets and rational expectations.

RANK with rational expectations A comparison of IT and AIT for the RE-RANK model without an ELB constraint already illustrates a key advantage of make-up monetary

³⁷We conduct 25 simulations with 40,000 periods each (excluding 1,000 additional burn-in periods). For each of the 25 simulations, we then calculate moments for variables of interest and report the average of these moments across simulations.



Figure 2: ELB episode under rational expectations

Notes: IRFs are for a large demand shock ($\varepsilon_{\beta,t} = 5\sigma_{\beta}$) and in percentage deviations from steady state.

	ELB incidence		Infla	Inflation (%)		put (%)
Policy rule	Freq. (%)	Avg. duration (quarters)	Avg.	Std.	Avg.	Std.
IT (w/o ELB)	0	0	2.00	0.59	0.00	0.89
AIT4 (w/o ELB)	0	0	2.00	0.57	0.00	0.91
AIT8 (w/o ELB)	0	0	2.00	0.57	0.00	0.92
IT	16.57	5.45	1.93	0.65	-0.07	1.11
AIT4	14.40	5.79	1.97	0.59	-0.05	1.07
AIT8	14.07	5.65	1.98	0.59	-0.05	1.07

Table 3: Results for HANK model with rational expectations (RE-HANK)

policy strategies emphasised in the literature: improved inflation stabilisation (see Table 2). Specifically, under AIT4, the standard deviation of inflation is about 10% (8 basis points) lower relative to IT. Extending the averaging window by an additional 4 years (16 model periods) further improves inflation stabilisation, but the additional improvement relative to AIT4 is only 1 basis point. The lower inflation volatility under AIT is achieved by successfully managing agents' expectations. If the average inflation target is missed in a given period, AIT prescribes that monetary policy will make up for this in the future. For instance, if average 4-year inflation is below target today, monetary policy is committed to be more expansionary in the subsequent periods. This commitment in turn drives up current expectations for inflation and real economic activity, providing additional stimulus today by encouraging households to increase consumption, firms to raise prices and unions to increase wages. Due to this "automatic stabiliser", all of this is accomplished without the need to move interest rates as much as in the IT case, which contributes to the improvement in price stabilisation. However, history dependence has the opposite impact on output volatility, which goes up as the length of the averaging window is increased. This highlights a potential downside of make-up strategies.³⁸

The commitment to make up for past misses is particularly valuable when the economy is at the ELB and monetary policy cannot offer additional stimulus by cutting current interest rates (see e.g. Eggertsson and Woodford, 2003). This can be illustrated by looking at impulse response functions for a large contractionary demand shock. This shock immediately pushes the policy rate to the ELB (see left panels in Figure 2), causing strong contractions in inflation and output. However, inflation drops substantially less during the ELB episode for AIT (solid blue and dashed red lines) relative to IT (dotted yellow line). Furthermore, under AIT, the policy rate does not stay at the ELB as long as under IT. These observations highlight the automatic stabiliser property of make-up strategies mentioned above. When average inflation is below target during an ELB episode, the AIT rule promises to keep interest rates "lower for longer", providing additional stimulus that the IT rule does not. Note that, although AIT generally promises to make up for past inflation misses, it only leads to

³⁸In contrast to the dampening impact on the volatility of inflation, we find that – in the absence of an ELB constraint – the qualitative effect of history dependence on output volatility depends on the degree of interest rate smoothing, ρ_R .

	Std. inflation (in %)			Std. output (in %)		
Cognitive ability n^*	IT	AIT4	AIT8	IT	AIT4	AIT8
0.5	0.64	0.64	0.64	1.26	1.27	1.28
1	0.68	0.68	0.68	0.96	1.00	1.02
2	0.74	0.72	0.74	0.89	0.95	0.98
3	0.77	0.72	0.74	0.88	0.95	0.98
6	0.78	0.70	0.69	0.88	0.95	0.97
∞	0.78	0.70	0.69	0.88	0.95	0.97

Table 4: Results for RANK model with reflective expectations (BR-RANK w/o ELB)

an actual inflation overshoot above target in Figure 2 for an averaging window of 8 years. In the stochastic model simulations, the lack of monetary stimulus at the ELB is reflected in a negative bias in inflation and output as well as heightened macroeconomic volatility. Consistent with Figure 2, these negative effects are mitigated by an AIT rule. It is worth noting that with an ELB constraint, inflation and output volatility are both lower under AIT, such that AIT leads to unambiguous improvements over IT in terms of macroeconomic stabilisation.

HANK with rational expectations Compared to the RE-RANK model, inflation and output do not behave much differently in the RE-HANK model – with or without the ELB (see Table 3). The only sizable difference involves a generally lower incidence and duration of ELB episodes in the incomplete markets case, as well as a generally lower degree of macroeconomic volatility. Importantly, the relative performance of AIT vs. IT does not change compared to the RE-RANK model, both qualitatively and quantitatively (see also right panels in Figure 2). These results likely reflect counteracting forces with respect to how market incompleteness affects the way AIT can manage economic activity by steering agents' expectations. On the one hand, the sensitivity of household consumption to promised future interest rate cuts is diminished by the presence of borrowing-constrained households (see McKay et al., 2016; Hagedorn et al., 2019). On the other hand, the marginal propensity to consume out of temporary income is substantially higher in the HANK model. Compared to the complete-markets case, this raises the importance of indirect monetary policy effects that stimulate economic activity via changes in disposable household income. For instance, redistributional effects that transfer wealth from low-MPC savers to high-MPC borrowers could in principle raise the potential of future interest rate cuts to provide economic stimulus (see e.g. Ferrante and Paustian, 2019). For our model, these counteracting forces appear to roughly cancel each other out, such that market incompleteness does not matter much for the effectiveness of make-up strategies.³⁹

³⁹As shown in Section 5.3, this finding is robust with respect to various assumptions made about the model, such as the denomination of debt and the cyclicality of household income risk.





(c) Inflation Bias (Annual, in %) -0.02 -0.04 -0.06 -0.08 0.5 1 2 3 6 ∞ Cognitive Ability







Figure 3: Results for RANK model with reflective expectations (BR-RANK)













Figure 4: Results for HANK model with reflective expectations (BR-HANK)

RANK with reflective expectations In contrast to market incompleteness, bounded rationality has a clearly visible impact on model predictions and the effectiveness of AIT. First consider the BR-RANK model without an ELB constraint (see Table 4).⁴⁰ Under IT, inflation tends to be more volatile than under AIT. Furthermore, cognitive ability n^* increases inflation volatility under IT, whereas n^* has a hump-shaped impact on inflation volatility under AIT. With respect to output, we observe that it is more stable when cognitive ability is high and becomes more volatile as the degree of history dependence goes up – in line with the findings for the RE-RANK model. The model properties are virtually the same for the BR-RANK model with $n^* = 6$ and the RE-RANK model (depicted in Table 4 as $n^* = \infty$), and are already close to the rational-expectations case for $n^* = 3$.⁴¹

When monetary policy faces an ELB constraint, inflation is more stable under an AIT rule (see Figure 3). Inflation volatility furthermore increases with cognitive ability under IT, whereas n^* again exhibits a hump-shaped impact on the standard deviation of inflation under AIT. By contrast, output is almost equally volatile under IT and AIT. In addition, output volatility is higher for low degrees of cognitive ability ($n^* = 0.5$) but hardly different between intermediate and high cognitive ability levels. Note that inflation and output tend to be only slightly less volatile under AIT8 than AIT4, illustrating decreasing returns to history dependence under AIT.

The better macroeconomic stabilisation properties of AIT are reflected in a lower incidence of ELB episodes. Importantly, by acting as an "automatic stabiliser", AIT reduces the first-order consequences of the ELB observed under IT, as measured by downward biases in average inflation and output, for all considered degrees of cognitive ability. However, the improvement vis-à-vis IT is quite small for degrees of cognitive ability that are in line with empirical and experimental evidence, i.e. $n^* \in \{0.5, 1, 2\}$.

HANK with reflective expectations Similar to the rational-expectations case, the behaviour of the HANK model economy is also close to that of the corresponding RANK model economy under reflective expectations (see Figure 4). This observation holds for both monetary policy strategies, and regardless of the agents' cognitive ability and the presence of an ELB constraint (see Table 5). This underscores our previous findings: Whereas market (in)completeness does not matter for the relative performance of AIT vs. IT in our model, the expectations process clearly does.

In addition, we do not find an interaction between market incompleteness and bounded rationality with respect to the attenuation of make-up strategies. Figure 5 illustrates this finding by plotting the inflation and output biases under AIT relative to IT. The relationship between the relative biases and cognitive ability n^* , given by the solid blue (AIT4)

 $^{^{40}}$ As under rational expectations, impulse responses for a demand-shock induced ELB episode paint a picture that is in line with the simulation-based results (see Figures 14 to 16 in Appendix C).

⁴¹This observation is reassuring from a computational perspective since we use a numerical equation solver for the rational-expectation model versions but the iterative procedure sketched in Appendix B for the cases with bounded rationality. As predicted by theory, both approaches yield the same results if n^* is sufficiently large.

	Std. inflation (in %)			Std. output (in %)		
Cognitive ability n^*	IT	AIT4	AIT8	IT	AIT4	AIT8
0.5	0.62	0.62	0.62	1.29	1.30	1.30
1	0.61	0.61	0.62	1.00	1.02	1.03
2	0.60	0.60	0.61	0.90	0.92	0.93
3	0.60	0.58	0.60	0.89	0.92	0.93
6	0.59	0.57	0.57	0.89	0.91	0.92
∞	0.59	0.57	0.57	0.89	0.91	0.92

Table 5: Results for HANK model with reflective expectations (BR-HANK w/o ELB)

and dashed red (AIT8) lines, is almost the same in the RANK model (left panels) and the HANK model (right panels). The only notable difference is a small upward shift of the lines in the HANK model, which reflects that make-up strategies are slightly less effective in lowering the biases under incomplete markets. A positive (negative) complementarity between market incompleteness and bounded rationality would require the lines to be steeper (flatter) for the HANK model, showing that bounded rationality dampens the effectiveness of make-up strategies more (less) in this case relative to the RANK model with complete markets. However, Figure 5 documents that the marginal impact of a change in n^* on the benefits of AIT is almost the same in the RANK and HANK models.

5.2 (Non-)Interaction between incomplete markets and bounded rationality

Farhi and Werning (2019) show that incomplete markets and bounded rationality can strongly attenuate the power of forward guidance when combined, but only have small attenuating effects when considered in isolation. In other words, there is a positive complementarity. Since make-up strategies operate via a built-in commitment to future interest rates, it might be surprising to see that our simulations do not indicate such a relationship between the incomplete markets and bounded rationality. In this section, we show how to reconcile our findings with those in Farhi and Werning (2019).

Forward guidance We start by performing a forward guidance experiment for different cognitive ability levels and conditional on market (in)completeness. Running this experiment allows us to check the existence of a complementarity between market incompleteness and bounded rationality with respect to forward guidance à la Farhi and Werning (2019). Given that we can indeed verify this property for our model, we can rule out differences between our model and the one used in Farhi and Werning (2019) as causes of a missing complementarity in the context of make-up strategies.⁴² We show the results of the experiment in Figures 6 and 7.

Figure 6 plots the period-t responses of output and inflation to a date-t announcement

⁴²Apart from small differences in terms of calibration, our model mainly differs from the quantitative stickyprice model employed in Farhi and Werning (2019) by featuring nominal wage stickiness, allowing for positive and nominal household debt as well as excluding outside liquidity.



Figure 5: Negative inflation and output biases for stochastic simulations

Notes: The biases are expressed as percentage deviations from the respective values under inflation targeting.

of a one-percent *real* rate cut in period $t + \tau$ (see McKay et al., 2016). First, consider the RANK model (left panels). Under rational expectations ($n^* = \infty$), the responsiveness of output does not depend on the horizon of the real rate cut realisation τ , whereas the inflation response exponentially increases with it. These properties constitute what is usually referred to as the "forward guidance puzzle" (see Del Negro et al., 2015; McKay et al., 2016), as they are difficult to square with economic intuition. By contrast, under reflective expectations, the output response declines with τ and inflation responds less strongly to real rate cuts further away in the future compared to $n^* = \infty$. What is more, the less sophisticated agents are, the more the output and inflation responses are attenuated. While the output response in the RANK model hardly changes with time horizon τ at all, the marginal effect of τ on the inflation response is declining for sufficiently low cognitive ability levels n^* . As a result, inflation remains bounded for $\tau \to \infty$ in this case. In that sense, bounded rationality can successfully solve the forward guidance puzzle, although its quantitative impact is not particularly large, which is in line with Farhi and Werning (2019).

In the HANK model (right panels), the output response declines with τ when agents'





Notes: The figure plots the percentage deviations of the respective variables from steady state upon announcement of a one-percent real rate cut in τ periods.

cognitive ability is sufficiently low, but increases with the horizon under rational expectations. As in the RANK model, the inflation response remains bounded for $\tau \to \infty$ when n^* is low, but not for $n^* = \infty$. Market incompleteness thus amplifies the forward guidance puzzle for our model calibration under rational expectations.⁴³ Note that forward guidance generally has stronger output and inflation effects in the HANK model relative to the RANK model if the time horizon τ is short, reflecting strong indirect effects of monetary policy that are are not (as) important under complete markets (see Kaplan et al., 2018).

To visualise the complementarity between market incompleteness and bounded rationality in the context of forward guidance, it is helpful to normalise the responses depicted in Figure 6 by the respective values under rational expectations. Given that the output and inflation responses in the RANK and HANK models differ in size, these normalised re-

⁴³As first pointed out by Werning (2015), this property depends on various model features, the cyclicality of household income risk and aggregate liquidity in particular. In our case, a redistribution of resources from savers (low MPCs) to borrowers (high MPCs) via inflation is crucial for making forward guidance more powerful under incomplete markets. When this redistribution of wealth is absent because bonds are indexed to inflation, the output effect declines with the horizon τ in all cases, including $n^* = \infty$.



Figure 7: Forward guidance experiment (normalised)

Notes: The figure plots the responses from Figure 6 but in percent of the respective responses under rational expectations $n^* = \infty$.

sponses illustrate the impact of bounded rationality on the power of forward guidance in a more transparent way.⁴⁴ Figure 7 plots these normalised responses for different cognitive ability levels and time horizons τ . By comparing these relative responses between RANK (left panels) and HANK (right panels), we observe that the attenuating effect of bounded rationality on the interest rate sensitivity of output and inflation is stronger under incomplete markets and increasing with time horizon τ .⁴⁵ In line with Farhi and Werning (2019), these differences between the RANK and HANK models reveal a positive complementarity between market incompleteness and bounded rationality with respect to the attenuation of forward guidance.

Discussion Forward guidance, as analysed in the previous section, is modelled as an unanticipated but perfectly observable news shock. As such, it constitutes a direct and uncondi-

⁴⁴Farhi and Werning (2019) construct and calibrate their model such that the output and inflation responses do not differ between the RANK and HANK models under rational expectations. This property makes it more straightforward to see how market incompleteness and bounded rationality interact.

⁴⁵Farhi and Werning (2019) refer to the former effect as *mitigation effect* and the latter one as *horizon effect*.

tional commitment to a future interest change, which agents are going to adjust their individual behaviour to. This particular behavioural response is the direct (or partial equilibrium) effect of the interest rate shock. In addition, the agents' behaviour reflects beliefs about the consequences of the interest rate shock for the future evolution of the aggregate economy. The corresponding response constitutes the indirect (or general equilibrium) effect of the interest rate change. The more sophisticated agents are, the better they understand these indirect effects and adjust their behaviour accordingly. Even the most naive agent (*n* close to 0) who does not adjust his expectations about the aggregate economy at all after the shock is announced, will, however, respond to the interest rate change itself. Bounded rationality therefore attenuates the indirect general equilibrium effects of the interest rate shock but not its direct effects, as agents can observe the interest rate change by assumption.

As shown by Farhi and Werning (2019) and verified for our model (see Figures 6 and 7), relaxing the rational-expectations assumption only mildly weakens the output effects of forward guidance under complete markets, even for quite large deviations from rational expectations. This finding reflects that in the RANK model, it is largely the direct effect of the interest rate change rather than its indirect effects that matter for the consumption response (see Kaplan et al., 2018; Hagedorn et al., 2019).⁴⁶ Given that the interest rate change itself is observed by households regardless of their cognitive ability n, this direct effect – and hence the dominant transmission channel of forward guidance under complete markets is not affected when the rational-expectations assumption is relaxed for the RANK model. By contrast, relaxing the complete-markets assumption by introducing occasionally-binding borrowing constraints weakens the direct effect of monetary policy. However, at the same time, it also raises the importance of indirect effects compared to the RANK case. A priori, it is therefore not obvious whether forward guidance is attenuated or even amplified in this case.⁴⁷ When combined, bounded rationality and market incompleteness jointly weaken direct and indirect effects, which yields a positive complementarity with respect to the attenuation of forward guidance.

In contrast to forward guidance, make-up strategies are rule-based and do not offer information about future interest rates that is directly observable to all types of agents. Instead, they only provide an indirect and conditional commitment to future interest rate changes. Specifically, whether monetary policy commits to adjust future interest rates depends on whether current average inflation meets the target or not. Future interest rate changes therefore have to be triggered first by a macroeconomic shock. While such a shock can be observed by the agents, whether and to what extent it will affect future interest rate movements has to be figured out by agents together with all other general equilibrium effects.

⁴⁶The direct effect of the interest rate change can further be decomposed into an income and a substitution effect. In the RANK model, it is primarily the latter that drives the size of the household consumption response and the aggregate output response as a result. This can be seen by looking at the real rate elasticity of output in the announcement period of a real rate shock. It is given as $(C/Y)\sigma^{-1}$ for the RE-RANK model, highlighting the tight link between the intertemporal elasticity of substitution σ^{-1} and the output response. The coefficient C/Y reflects that output does not equal consumption in the steady state if there is a fixed production $\cos \Phi > 0$.

⁴⁷As discussed above, our quantitative model implies that forward guidance is more effective in the HANK model for $n^* = \infty$.

In contrast to the forward guidance experiment, the transmission of monetary policy under an interest rate rule hence operates entirely via indirect general equilibrium effects.⁴⁸ Given that these effects are not directly observable to the agents, bounded rationality has a symmetric impact on how make-up strategies affect the economy under complete and incomplete markets. A complementarity as observed for forward guidance is therefore absent (or too small to notice), as it relies on market incompleteness and bounded rationality attenuating different types of effects.⁴⁹

5.3 Sensitivity

In this section, we assess the sensitivity of our simulation results with respect to different model assumptions. First, we allow for real instead of nominal household debt. Second, we consider a model calibration where goods prices and nominal wages are partially indexed to lagged inflation. Third, we allow household income risk to change endogenously with macroeconomic conditions. Fourth, we consider model versions where only one type of forward-looking agent is boundedly rational. Fifth, we investigate the role of interest rate smoothing for our results.

Debt denomination When household debt is nominal, a deflationary shock redistributes resources from high-MPC borrowers to low-MPC savers. This transfer of wealth increases the severity of ELB episodes and raises the benefits of policies that attenuate deflationary or disinflationary spirals. When debt is real, this (surprise) redistribution is absent and one would expect the benefits of make-up strategies to be lower compared to a scenario with nominal household debt (see Table 3). While we confirm this intuition (see Table 6 in Appendix C), the quantitative differences between RE-HANK model versions with nominal and real debt are of negligible size. The relative performance of IT and AIT thus is not affected by the denomination of household debt.

Price and wage indexation For our baseline calibration, we assumed price- and wagesetting to be entirely forward-looking, i.e. $\iota_p = \iota_w = 1$. We now assume that prices and wages of non-resetting firms and unions are partially indexed to lagged inflation ($\iota_p = \iota_w = 0.9$).⁵⁰ This assumption introduces inertia into price- and wage-setting behaviour and thereby reduces the relative importance of expected price and wage inflation for the economy. Although one might expect AIT to lose some of its stabilising power due to an

⁴⁸Only the aggregate shocks themselves have a direct effect on the agents' behaviour in this case.

⁴⁹In line with results presented by Bianchi-Vimercati et al. (2021) for unconventional fiscal policies in a RANK model, discretionary and perfectly observable interest rate cuts are more effective at stimulating an economy when agents are boundedly rational agents than rule-based tax cuts. Closely related, Angeletos and Sastry (2021) emphasise the benefits of communicating monetary policy by specifying targets for variables of interest (like inflation or output) rather than instruments (like interest rates) when the public faces cognitive constraints.

⁵⁰For ι_p -values below 0.9, we found that for some model versions, there is no equilibrium when the economy hits the ELB during some simulations, as identified by Holden (2016)'s method (see Section B.2). To compare all relevant model versions, we therefore took the lowest ι_p -value to ensure that this issue did not occur for any of the considered model versions.

impaired ability to manage agents' expectations, the relative performance of IT and AIT does not change (see Tables 7 and 8 in Appendix C).

Cyclical income risk The literature has emphasised the importance of cyclical household income risk for monetary policy transmission in general and forward guidance in particular (see e.g. Werning, 2015; Acharya and Dogra, 2020). Specifically, whereas countercyclical income risk strengthens the power of forward guidance in a HANK context compared to the complete markets case, procyclical income risk attenuates it. The intuition is straightforward. Countercyclical (procyclical) income risk raises (lowers) MPCs in response to an expansionary (contractionary) monetary policy, which amplifies (dampens) the economy's response by strengthening (weakening) indirect effects. To allow income risk to systematically change with aggregate economic conditions in our model, we let individual household productivity depend on aggregate output by adopting a reduced form relationship as in Auclert and Rognlie (2020).⁵¹ In particular, we assume that individual productivity is given as $z_i \Gamma(z_i, Y_t)$, with

$$\Gamma(z_i, Y_t) = \frac{z_i^{\zeta \log(Y_t/Y)}}{\sum_j \Pr(z_j) z_j^{1+\zeta \log(Y_t/Y)}}$$

This functional form implies that the cross-sectional standard deviation of log-household productivity is $[1 + \zeta \log(Y_t/Y)] \times \text{SD}(\log z_i)$. Parameter ζ directly governs the cyclicality of idiosyncratic income risk. In the previous sections we assumed $\zeta = 0$, implying acyclical income risk. For $\zeta < 0$, the model features countercyclical income risk, whereas income risk becomes procyclical for $\zeta > 0$.

Tables 9 and 10 in Appendix C show simulation statistics for RE-HANK model versions with $\zeta = -1$ and $\zeta = 1$, respectively. As expected, countercyclical income risk raises macroeconomic volatility in general and results in stronger inflation and output biases. The latter reflects stronger contractions at the aggregate level during ELB episodes compared to $\zeta = 0$ (see Table 3). The opposite observations can be made when income risk is procyclical. Importantly, these differences relative to the case with acyclical income risk apply uniformly across policy rules. The relative performance of make-up strategies is therefore not affected by the cyclicality of income risk.

Heterogeneous cognitive ability So far, we assumed that households, intermediate-good firms and labour unions all possess the same degree of cognitive ability. Because of this assumption, one cannot tell which of the three types of agents matters most for the impact of bounded rationality on the effectiveness of make-up strategies.⁵² To clarify this issue, we solve and simulate our model under the assumption that only one of the three aforementioned agents is non-rational. Specifically, we assume that this particular type of agent is

⁵¹Bayer et al. (2020) model the cyclicality of income risk in a reduced-form way as well. Their model estimation implies a positive impact of aggregate income on individual income risk.

⁵²Moreover, a recent empirical literature on subjective expectations shows that different types of agents possess different degrees of rationality (see e.g. Weber et al., 2022).

characterised by n = 2, while the other ones are perfectly rational. As argued in Section 3, a value of 2 is the highest cognitive ability level *n* that can be considered in line with micro-evidence.⁵³ Figures 17 and 18 in Appendix C show that the model statistics slightly vary depending on which type of agent is assumed to be non-rational. However, the size of these differences is quantitatively small. Bounded rationality therefore has a quite uniform impact on monetary policy transmission under all three policy strategies. The results of this paper thus do not depend on which type of agent is boundedly rational.

Interest rate smoothing While an interest rate rule with interest rate smoothing is standard in the literature, it could potentially distort the comparison between IT and AIT because under interest rate smoothing even IT features a history-dependent element. We therefore simulate the model also without interest rate smoothing ($\rho_R = 0$) to assess the robustness of our findings along this dimension. To enhance comparability, we recalibrate the demand shock volatility relative to Section 3 to again match an ELB frequency of 20% for the RE-RANK case. The resulting value is $\sigma_{\beta} = 0.0047$. As can be seen in Figures 19 and 20 in Appendix **C**, the main findings of this paper (see Section 5.1) do not change once interest rate smoothing is abandoned.

6 Conclusion

In this paper, we have studied to what extent the effectiveness of make-up strategies depends on the assumptions of rational expectations and complete markets. Specifically, we have performed simulations for a HANK model that was extended to allow for non-rational expectation formation and an occasionally-binding ELB constraint. Our simulations suggest that the power of make-up strategies does not particularly depend on whether markets are complete. By contrast, if agents are quite limited in their ability to process the macroeconomic consequences of aggregate shocks, the effectiveness of make-up strategies suffers, which can make them almost as useful as standard inflation targeting for addressing the adverse effects of an ELB. Since we observe that, for a given level of cognitive ability, the relative performance of AIT and IT hardly differs between our HANK model and a corresponding RANK model, one can argue that (bounded) rationality matters more for the effectiveness of make-up policies than market (in)completeness. This observation also implies that market incompleteness and bounded rationality do not complement each other in attenuating the benefits of make-up strategies.

⁵³This value is also the lowest one where we did not encounter non-existence of equilibrium during some ELB episodes for any of the considered model versions.

References

- ACHARYA, S. AND K. DOGRA (2020): "Understanding HANK: Insights From a PRANK," *Econometrica*, 88, 1113–1158.
- AIYAGARI, S. (1994): "Uninsured Idiosyncratic Risk and Aggregate Saving," *Quarterly Journal of Economics*, 109, 659–684.
- ANDRADE, P., J. GALÍ, H. LE BIHAN, AND J. MATHERON (2018): "The Optimal Inflation Target and the Natural Rate of Interest," NBER Working Paper No. 24328.
- ANGELETOS, G.-M. AND C. LIAN (2018): "Forward Guidance without Common Knowledge," *American Economic Review*, 108, 2477–2512.
- ANGELETOS, G.-M. AND K. SASTRY (2021): "Managing Expectations: Instruments vs. Targets," *Quarterly Journal of Economics*, 136, 2467–2532.
- ARAD, A. AND A. RUBINSTEIN (2012): "The 11-20 Money Request Game: Evaluating the Upper Bound of Level-k Thinking," *American Economic Review*, 102, 3561–3573.
- ARIAS, J., M. BODENSTEIN, H. CHUNG, T. DRAUTZBURG, AND A. RAFFO (2020): "Alternative Strategies: How Do They Work? How Might They Help?" Finance and Economics Discussion Series 2020-068.
- AUCLERT, A., B. BARDÓCZY, M. ROGNLIE, AND L. STRAUB (2021): "Using the Sequence-Space Jacobian to Solve and Estimate Heterogeneous-Agent Models," *Econometrica*, 89, 2375–2408.
- AUCLERT, A. AND M. ROGNLIE (2020): "Inequality and Aggregate Demand," Mimeo.
- AUCLERT, A., M. ROGNLIE, AND L. STRAUB (2020): "Micro Jumps, Macro Humps: Monetary Policy and Business Cycles in an Estimated HANK Model," NBER Working Paper No. 26647.
- BAYER, C., B. BORN, AND R. LUETTICKE (2020): "Shocks, Frictions, and Inequality in US Business Cycles," Mimeo.
- BAYER, C. AND R. LUETTICKE (2020): "Solving Discrete Time Heterogeneous Agent Models with Aggregate Risk and Many Idiosyncratic States by Perturbation," *Quantitative Economics*, 11, 1253–1288.
- BAYER, C., R. LUETTICKE, L. PHAM-DAO, AND V. TJADEN (2019): "Precautionary Savings, Illiquid Assets, and the Aggregate Consequences of Shocks to Household Income Risk," *Econometrica*, 87, 255–290.
- BERGMAN, N., D. MATSA, AND M. WEBER (2022): "Inclusive Monetary Policy: How Tight Labor Markets Facilitate Broad-Based Employment Growth," NBER Working Paper No. 29651.

- BERNANKE, B. (2020): "The New Tools of Monetary Policy," *American Economic Review*, 110, 943–983.
- BERSSON, B., P. HÜRTGEN, AND M. PAUSTIAN (2019): "Expectations Formation, Sticky Prices, and the ZLB," Deutsche Bundesbank Discussion Paper No. 34/2019.
- BHANDARI, A., D. EVANS, M. GOLOSOV, AND T. SARGENT (2021): "Inequality, Business Cycles, and Monetary-Fiscal Policy," *Econometrica*, 89, 2559–2599.
- BIANCHI-VIMERCATI, R. (2022): "Learning Unconventional Policies: Forward Guidance with Integrated Reasoning," Mimeo.
- BIANCHI-VIMERCATI, R., M. EICHENBAUM, AND J. GUERREIRO (2021): "Fiscal Policy at the Zero Lower Bound without Rational Expectations," NBER Working Paper No. 29134.
- BILBIIE, F. (2021): "Monetary Policy and Heterogeneity: An Analytical Framework," Mimeo.
- BLANCO, A. (2021): "Optimal Inflation Target in an Economy with Menu Costs and a Zero Lower Bound," American Economic Journal: Macroeconomics, 13, 108–141.
- BODENSTEIN, M., C. ERCEG, AND L. GUERRIERI (2009): "The Effects of Foreign Shocks when Interest Rates Are at Zero," International Finance Discussion Papers 983.
- BODENSTEIN, M., L. GUERRIERI, AND C. GUST (2013): "Oil Shocks and the Zero Bound on Nominal Interest Rates," *Journal of International Money and Finance*, 32, 941–967.
- BODENSTEIN, M., J. HEBDEN, AND F. WINKLER (2022): "Learning and Misperception of Makeup Strategies," *Journal of Economic Dynamics and Control*, 139, 104417.
- BOPPART, T., P. KRUSELL, AND K. MITMAN (2018): "Exploiting MIT Shocks in Heterogeneous-Agent Economies: The Impulse Response as a Numerical Derivative," *Journal of Economic Dynamics and Control*, 89, 68–92.
- BRAYTON, F., A. LEVIN, R. LYON, AND J. WILLIAMS (1997): "The Evolution of Macro Models at the Federal Reserve Board," *Carnegie-Rochester Conference Series on Public Policy*, 47, 43–81.
- BUDIANTO, F., T. NAKATA, AND S. SCHMIDT (2020): "Average Inflation Targeting and the Interest Rate Lower Bound," ECB Working Paper Series No. 2394.
- CARLSTROM, C., T. FUERST, AND M. PAUSTIAN (2015): "Inflation and Output in New Keynesian Models with a Transient Interest Rate Peg," *Journal of Monetary Economics*, 76, 230–243.
- CARROLL, C. (2006): "The Method of Endogenous Gridpoints for Solving Dynamic Stochastic Optimization Problems," *Economics Letters*, 91, 312–320.

- CHEN, H., V. CÚRDIA, AND A. FERRERO (2012): "The Macroeconomic Effects of Large-Scale Asset Purchase Programmes," *Economic Journal*, 122, F289–315.
- CHETTY, R., A. GUREN, D. MANOLI, AND A. WEBER (2011): "Are Micro and Macro Labor Supply Elasticities Consistent? A Review of Evidence on the Intensive and Extensive Margins," *American Economic Review*, 101, 471–475.
- CHRISTIANO, L., M. EICHENBAUM, AND M. TRABANDT (2016): "Unemployment and Business Cycles," *Econometrica*, 84, 1523–1569.
- COIBION, O., Y. GORODNICHENKO, S. KUMAR, AND J. RYNGAERT (2021): "Do You Know That I Know That You Know...? Higher-Order Beliefs in Survey Data," *Quarterly Journal of Economics*, 136, 1387–1446.
- COIBION, O., Y. GORODNICHENKO, AND J. WIELAND (2012): "The Optimal Inflation Rate in New Keynesian Models: Should Central Banks Raise Their Inflation Targets in Light of the Zero Lower Bound?" *Review of Economic Studies*, 79, 1371–1406.
- COSTA-GOMES, M. AND V. CRAWFORD (2006): "Cognition and Behavior in Two-Person Guessing Games: An Experimental Study," *American Economic Review*, 96, 1737–1768.
- DEL NEGRO, M., M. GIANNONI, AND C. PATTERSON (2015): "The Forward Guidance Puzzle," Federal Reserve Bank of New York Staff Reports No. 574.
- DJEUTEM, E., M. HE, A. REZA, AND Y. ZHANG (2022): "Household Heterogeneity and the Performance of Monetary Policy Frameworks," Bank of Canada Staff Working Papers 22-12.
- DOBREW, M., R. GERKE, S. GIESEN, AND J. RÖTTGER (2021): "A Comparison of Monetary Policy Rules in a HANK Model," Deutsche Bundesbank Technical Paper 02/2021.
- DUPRAZ, S., H. LE BIHAN, AND J. MATHERON (2022): "Make-Up Strategies with Finite Planning Horizons but Forward-Looking Asset Prices," Banque de France Working Paper No. 862.
- EGGERTSSON, G. AND M. WOODFORD (2003): "The Zero Bound on Interest Rates and Optimal Monetary Policy," *Brookings Papers on Economic Activity*, 34, 139–211.
- ERCEG, C., D. HENDERSON, AND A. LEVIN (2000): "Optimal Monetary Policy with Staggered Wage and Price Contracts," *Journal of Monetary Economics*, 46, 281–313.
- ERCEG, C., Z. JAKAB, AND J. LINDÉ (2021): "Monetary Policy Strategies for the European Central Bank," *Journal of Economic Dynamics and Control*, 132, 104211.
- EUSEPI, S. AND B. PRESTON (2018): "The Science of Monetary Policy: An Imperfect Knowledge Perspective," *Journal of Economic Literature*, 56, 3–59.

- EVANS, G., S. HONKAPOHJA, AND K. MITRA (2022): "Expectations, Stagnation and Fiscal Policy: a Nonlinear Analysis," *International Economic Review*, 63, 1397–1425.
- FAIR, R. AND J. TAYLOR (1983): "Solution and Maximum Likelihood Estimation of Dynamic Nonlinear Rational Expectations Models," *Econometrica*, 51, 1169–1185.
- FARHI, E. AND I. WERNING (2019): "Monetary Policy, Bounded Rationality, and Incomplete Markets," *American Economic Review*, 109, 3887–3928.
- FEIVESON, L., N. GOERNEMANN, J. HOTCHKISS, K. MERTENS, AND J. SIM (2020): "Distributional Considerations for Monetary Policy Strategy," Finance and Economics Discussion Series 2020-073.
- FERNÁNDEZ-VILLAVERDE, J., J. MARBET, G. NUÑO, AND O. RACHEDI (2021): "Inequality and the Zero Lower Bound," Mimeo.
- FERRANTE, F. AND M. PAUSTIAN (2019): "Household Debt and the Heterogeneous Effects of Forward Guidance," International Finance Discussion Papers 1267.
- GABAIX, X. (2020): "A Behavioral New Keynesian Model," *American Economic Review*, 110, 2271–2327.
- GALÍ, J. (2015): Monetary Policy, Inflation, and the Business Cycle: An Introduction to the New Keynesian Framework, Princeton, NJ: Princeton University Press.
- GARCÍA-SCHMIDT, M. AND M. WOODFORD (2015): "Are Low Interest Rates Deflationary? A Paradox of Perfect-Foresight Analysis," NBER Working Paper No. 21614.
- (2019): "Are Low Interest Rates Deflationary? A Paradox of Perfect-Foresight Analysis," *American Economic Review*, 109, 86–120.
- GERKE, R., S. GIESEN, AND D. KIENZLER (2020): "Interest Rate Pegs and the Reversal Puzzle: On the Role of Anticipation," Deutsche Bundesbank Discussion Paper No. 50/2020.
- GORNEMANN, N., K. KUESTER, AND M. NAKAJIMA (2021): "Doves for the Rich, Hawks for the Poor? Distributional Consequences of Systematic Monetary Policy," Mimeo.
- GRANDMONT, J.-M. (1977): "Temporary General Equilibrium Theory," *Econometrica*, 45, 535–572.
- GRIMAUD, A. (2021): "Precautionary Saving and Un-Anchored Expectations," Mimeo.
- GUERRIERI, L. AND M. IACOVIELLO (2015): "OccBin: A Toolkit for Solving Dynamic Models with Occasionally Binding Constraints Easily," *Journal of Monetary Economics*, 70, 22–38.

- HAGEDORN, M., J. LUO, I. MANOVSKII, AND K. MITMAN (2019): "Forward Guidance," Journal of Monetary Economics, 102, 1–23.
- HEER, B. AND A. MAUSSNER (2009): Dynamic General Equilibrium Modelling: Computational Methods and Applications, Berlin: Springer.
- HICKS, J. (1939): Value and Capital, Oxford: Oxford University Press.
- HOLDEN, T. (2016): "Computation of Solutions to Dynamic Models with Occasionally Binding Constraints," Mimeo.
- HOLDEN, T. AND M. PAETZ (2012): "Efficient Simulation of DSGE Models with Inequality Constraints," University of Surrey Discussion Paper 15/12.
- HONKAPOHJA, S. AND K. MITRA (2020): "Price Level Targeting with Evolving Credibility," *Journal of Monetary Economics*, 116, 88–103.
- HUGGETT, M. (1993): "The Risk-Free Rate in Heterogeneous-Agent Incomplete-Insurance Economies," *Journal of Economic Dynamics and Control*, 17, 953–969.
- İMROHOROĞLU, A. (1992): "The Welfare Cost of Inflation under Imperfect Insurance," Journal of Economic Dynamics and Control, 16, 79–91.
- IOVINO, L. AND D. SERGEYEV (2021): "Central Bank Balance Sheet Policies Without Rational Expectations," Mimeo.
- JEENAS, P. (2020): "Firm Balance Sheet Liquidity, Monetary Policy Shocks, and Investment Dynamics," Mimeo.
- JUNGHERR, J., M. MEIER, T. REINELT, AND I. SCHOTT (2021): "Corporate Debt Maturity Matters For Monetary Policy," Mimeo.
- KAPLAN, G., B. MOLL, AND G. VIOLANTE (2018): "Monetary Policy According to HANK," *American Economic Review*, 108, 697–743.
- KAPLAN, G. AND G. VIOLANTE (2018): "Microeconomic Heterogeneity and Macroeconomic Shocks," *Journal of Economic Perspectives*, 32, 167–94.
- KAPLAN, G., G. VIOLANTE, AND J. WEIDNER (2014): "The Wealthy Hand-to-Mouth," *Brookings Papers on Economic Activity*, 45, 77–138.
- KNEELAND, T. (2015): "Identifying Higher-Order Rationality," *Econometrica*, 83, 2065– 2079.
- KUHN, M. AND J.-V. RÍOS-RULL (2016): "2013 Update on the US Earnings, Income, and Wealth Distributional Facts: A View from Macroeconomics," *Federal Reserve Bank of Minneapolis Quarterly Review*, 37, 2–73.

- LASÉEN, S. AND L. SVENSSON (2011): "Anticipated Alternative Policy Rate Paths in Policy Simulations," *International Journal of Central Banking*, 7, 1–35.
- LEE, D. (2021): "Quantitative Easing and Inequality," Mimeo.
- LINDAHL, E. (1939): Theory of Money and Capital, London: Allen and Unwin.
- LUETTICKE, R. (2021): "Transmission of Monetary Policy with Heterogeneity in Household Portfolios," *American Economic Journal: Macroeconomics*, 13, 1–25.
- MCKAY, A., E. NAKAMURA, AND J. STEINSSON (2016): "The Power of Forward Guidance Revisited," *American Economic Review*, 106, 3133–3158.
- MCKAY, A. AND R. REIS (2016): "The Role of Automatic Stabilizers in the U.S. Business Cycle," *Econometrica*, 84, 141–194.
- MCKAY, A. AND J. WIELAND (2021): "Lumpy Durable Consumption Demand and the Limited Ammunition of Monetary Policy," *Econometrica*, 89, 2717–2749.
- MEH, C., J.-V. RÍOS-RULL, AND Y. TERAJIMA (2010): "Aggregate and Welfare Effects of Redistribution of Wealth under Inflation and Price-Level Targeting," *Journal of Monetary Economics*, 57, 637–652.
- MELE, A., K. MOLNÁR, AND S. SANTORO (2020): "On the Perils of Stabilizing Prices When Agents Are Learning," *Journal of Monetary Economics*, 115, 339–353.
- MOLAVI, P. (2022): "Simple Models and Biased Forecasts," Mimeo.
- NAGEL, R. (1995): "Unraveling in Guessing Games: An Experimental Study," American Economic Review, 85, 1313–1326.
- NAKATA, T. AND S. SCHMIDT (2019): "Gradualism and Liquidity Traps," *Review of Economic Dynamics*, 31, 182–199.
- OTTONELLO, P. AND T. WINBERRY (2020): "Financial Heterogeneity and the Investment Channel of Monetary Policy," *Econometrica*, 88, 2473–2502.
- PFÄUTI, O. AND F. SEYRICH (2022): "A Behavioral Heterogeneous Agent New Keynesian Model," Mimeo.
- PRESTON, B. (2005): "Learning About Monetary Policy Rules When Long-Horizon Expectations Matter," *International Journal of Central Banking*, 1, 81–126.
- QIU, Z. (2018): "Level-k DSGE and Monetary Policy," Mimeo.
- REITER, M. (2009): "Solving Heterogeneous-Agent Models by Projection and Perturbation," *Journal of Economic Dynamics and Control*, 33, 649–665.

- (2018): "Comments on "Exploiting MIT Shocks in Heterogeneous-Agent Economies: The Impulse Response as a Numerical Derivative" by T. Boppart, P. Krusell and K. Mitman," *Journal of Economic Dynamics and Control*, 89, 93–99.
- SCHAAB, A. (2020): "Micro and Macro Uncertainty," Mimeo.
- SCHMITT-GROHÉ, S. AND M. URIBE (2004): "Solving Dynamic General Equilibrium Models Using a Second-Order Approximation to the Policy Function," *Journal of Economic Dynamics and Control*, 28, 755–775.
- ——— (2006): "Comparing Two Variants of Calvo-Type Wage Stickiness," NBER Working Paper No. 12740.
- SVENSSON, L. (1999): "Price-Level Targeting versus Inflation Targeting: A Free Lunch?" Journal of Money, Credit, and Banking, 31, 277–295.
- TAYLOR, J. (1993): "Discretion versus Policy Rules in Practice," Carnegie-Rochester Conference Series on Public Policy, 39, 195–214.
- (1999): "A Historical Analysis of Monetary Policy Rules," in *Monetary Policy Rules*, ed. by J. Taylor, Chicago, IL: University of Chicago Press, 319–348.
- VESTIN, D. (2006): "Price-Level versus Inflation Targeting," Journal of Monetary Economics, 53, 1361–1376.
- VISSING-JØRGENSEN, A. (2002): "Limited Asset Market Participation and the Elasticity of Intertemporal Substitution," *Journal of Political Economy*, 110, 825–853.
- WEBER, M., F. D'ACUNTO, Y. GORODNICHENKO, AND O. COIBION (2022): "The Subjective Inflation Expectations of Households and Firms: Measurement, Determinants, and Implications," NBER Working Paper No. 30046.
- WERNING, I. (2015): "Incomplete Markets and Aggregate Demand," NBER Working Paper No. 21448.
- WOODFORD, M. (2013): "Macroeconomic Analysis Without the Rational Expectations Hypothesis," *Annual Review of Economics*, 5, 303–346.

(2018): "Monetary Policy Analysis When Planning Horizons Are Finite," in *NBER Macroeconomics Annual*, ed. by M. Eichenbaum and J. Parker, Cambridge, MA: MIT Press, vol. 33, 1–50.

- WORK STREAM ON THE PRICE STABILITY OBJECTIVE (2021): "The ECB's Price Stability Framework: Past Experience, and Current and Future Challenges," Occasional Paper Series 269, European Central Bank.
- YOUNG, E. (2010): "Solving the Incomplete Markets Model with Aggregate Uncertainty Using the Krusell-Smith Algorithm and Non-Stochastic Simulations," *Journal of Economic Dynamics and Control*, 34, 36–41.

A Appendix: Derivation of optimal decision rules

This section derives the optimal decision rules shown in Section 2.1.

Households Define income $y_t = w_t N_t + d_t$, which is taken as given by all households. The household period budget constraint is

$$b_{i,t-1} \frac{R_{t-1}}{\Pi_t} + y_t = \tilde{b}_{i,t} + \tilde{c}_{i,t},$$

for current period t and

$$\tilde{b}_{i,t+s-1} \frac{R^e_{t+s-1}}{\Pi^e_{t+s}} + y^e_{t+s} = \tilde{b}_{i,t+s} + \tilde{c}_{i,t+s}$$

for future periods t + s, s > 0.

Repeated forward substitution of (planned) end-of-period bonds via

$$\tilde{b}_{i,t+s-1} = \left(\tilde{b}_{i,t+s} + \tilde{c}_{i,t+s} - y_{t+s}^e\right) \left(\frac{R_{t+s-1}^e}{\Pi_{t+s}^e}\right)^{-1},$$

yields the intertemporal budget constraint,

$$b_{i,t-1}\frac{R_{t-1}}{\Pi_t} = \sum_{s=0}^{\infty} \left(\prod_{u=0}^{s} \left(\frac{R_{t+u}^e}{\Pi_{t+u+1}^e} \right)^{-1} \right) (c_{i,t+s} - y_{t+s}),$$

or

$$\tilde{c}_{i,t} + \sum_{s=1}^{\infty} \left(\prod_{u=1}^{s} \left(\frac{R_{t+u-1}^{e}}{\Pi_{t+u}^{e}} \right)^{-1} \right) \tilde{c}_{i,t+s} = b_{i,t-1} \frac{R_{t-1}}{\Pi_{t}} + y_{t} + \sum_{s=1}^{\infty} \left(\prod_{u=1}^{s} \left(\frac{R_{t+u-1}^{e}}{\Pi_{t+u}^{e}} \right)^{-1} \right) y_{t+s}^{e}.$$

Household *i* maximises expected lifetime utility,

$$\sum_{s=0}^{\infty} \left(\prod_{k=0}^{s} \beta_{t+k-1} \right) \left\{ u\left(\tilde{c}_{i,t+s} \right) - v\left(L_{t+s}^{e} \right) \right\},$$

by choosing consumption sequence $\{\tilde{c}_{i,t+s}\}_{s=0}^{\infty}$ subject to the intertemporal budget constraint. With felicity function $u(c) = (c^{1-\sigma} - 1)/(1-\sigma)$, the first-order condition for (planned) consumption in period t + s is

$$\tilde{c}_{i,t}^{-\sigma} = \tilde{c}_{i,t+s}^{-\sigma} \prod_{u=1}^{s} \beta_{t+u-1} \frac{R_{t+u-1}^e}{\prod_{t+u}^e},$$

or

$$\tilde{c}_{i,t+s} = \tilde{c}_{i,t} \left(\prod_{u=1}^{s} \beta_{t+u-1} \frac{R_{t+u-1}^{e}}{\prod_{t+u}^{e}} \right)^{1/\sigma}.$$

By using this condition to replace $\tilde{c}_{i,t+s}$ in the intertemporal household budget con-

straint, we obtain

$$\tilde{c}_{i,t} + \tilde{c}_{i,t} \sum_{s=1}^{\infty} \left(\prod_{u=1}^{s} \left(\frac{R_{t+u-1}^{e}}{\Pi_{t+u}^{e}} \right)^{-1} \right) \left(\prod_{u=1}^{s} \beta_{t+u-1} \frac{R_{t+u-1}^{e}}{\Pi_{t+u}^{e}} \right)^{1/\sigma} \\ = b_{i,t-1} \frac{R_{t-1}}{\Pi_{t}} + y_{t} + \sum_{s=1}^{\infty} \left(\prod_{u=1}^{s} \left(\frac{R_{t+u-1}^{e}}{\Pi_{t+u}^{e}} \right)^{-1} \right) y_{t+s}^{e}.$$

Solving for $\tilde{c}_{i,t}$ then yields

$$\tilde{c}_{i,t} = \frac{b_{i,t-1} \frac{R_{t-1}}{\Pi_t} + y_t + \sum_{s=1}^{\infty} \left(\prod_{u=1}^{s} \left(\frac{R_{t+u-1}^e}{\Pi_{t+u}^e} \right)^{-1} \right) y_{t+s}^e}{1 + \sum_{s=1}^{\infty} \left(\prod_{u=1}^{s} \left(\frac{R_{t+u-1}^e}{\Pi_{t+u}^e} \right)^{-1} \right) \left(\prod_{u=1}^{s} \beta_{t+u-1} \frac{R_{t+u-1}^e}{\Pi_{t+u}^e} \right)^{1/\sigma}},$$

or simplified

$$\tilde{c}_{i,t} = \frac{b_{i,t-1} \frac{R_{t-1}}{\Pi_t} + y_t + \sum_{s=1}^{\infty} \left(\prod_{u=1}^{s} \left(\frac{R_{t+u-1}^e}{\Pi_{t+u}^e} \right)^{-1} \right) y_{t+s}^e}{1 + \sum_{s=1}^{\infty} \prod_{u=1}^{s} \beta_{t+u-1}^{1/\sigma} \left(\frac{R_{t+u-1}^e}{\Pi_{t+u}^e} \right)^{1/\sigma-1}}.$$

Intermediate-good firms In period *t*, intermediate-good firm *j* sets price $P_{j,t}$ to maximise the expected present value of its profits,

$$\sum_{s=0}^{\infty} (\beta \xi_p)^s \left[\begin{array}{c} (P_t^*)^{1-\theta_p} \left(\frac{\prod_{u=1}^s \Pi^{i_p} (\Pi_{t+u-1}^e)^{1-i_p}}{P_{t+s}^e} \right)^{1-\theta_p} Y_{t+s}^e \\ -\lambda_{t+s} w_{t+s}^e (P_t^*)^{-\theta_p} \left(\frac{\prod_{u=1}^s \Pi^{i_p} (\Pi_{t+u-1}^e)^{1-i_p}}{P_{t+s}^e} \right)^{-\theta_p} Y_{t+s}^e - \Phi \end{array} \right],$$

where we used that $Y_{j,t+s} = (P_{j,t+s}/P_{t+s})^{-\theta_p} Y_{t+s}$ and $P_{j,t+s} = P_{j,t} \times \prod_{u=1}^{s} \left[\prod_{u=1}^{t_p} (\prod_{t+u-1}^{e})^{1-t_p} \right].$

Let P_t^* denote the optimal reset price, which is the same across all firms *j* that can adjust prices in period *t*. The first-order condition for P_t^* can be written as

$$P_{t}^{*} = \frac{\theta_{p}}{\theta_{p} - 1} \frac{\sum\limits_{s=0}^{\infty} (\beta \xi_{p})^{s} \left(\frac{\prod_{u=1}^{s} \Pi^{\iota_{p}} (\Pi_{t+u-1}^{e})^{1-\iota_{p}}}{P_{t+s}^{e}}\right)^{-\theta_{p}} \lambda_{t+s} w_{t+s}^{e} Y_{t+s}^{e}}{\sum\limits_{s=0}^{\infty} (\beta \xi_{p})^{s} \left(\frac{\prod_{u=1}^{s} \Pi^{\iota_{p}} (\Pi_{t+u-1}^{e})^{1-\iota_{p}}}{P_{t+s}^{e}}\right)^{1-\theta_{p}} Y_{t+s}^{e}}.$$

Define optimal reset inflation as $\Pi_{t+s}^* = P_{t+s}^*/P_{t+s-1}$. Using $P_{t+s}^e = P_t \times \prod_{u=1}^s \Pi_{t+u}^e$ and $\Pi_{t+s} = P_{t+s}/P_{t+s-1}$, we can write the expression above as

$$\Pi_{t}^{*} = \frac{\theta_{p}}{\theta_{p}-1} \frac{\sum_{s=0}^{\infty} (\beta \xi_{p})^{s} \left(\prod_{u=1}^{s} \frac{\Pi^{l_{p}} (\Pi_{t+u-1}^{e})^{1-l_{p}}}{\Pi_{t+u}^{e}} \right)^{-\theta_{p}} \lambda_{t+s} w_{t+s}^{e} Y_{t+s}^{e}}{\sum_{s=0}^{\infty} (\beta \xi_{p})^{s} \left(\prod_{u=1}^{s} \frac{\Pi^{l_{p}} (\Pi_{t+u-1}^{e})^{1-l_{p}}}{\Pi_{t+u}^{e}} \right)^{1-\theta_{p}} Y_{t+s}^{e}} \Pi_{t}.$$

Labour unions Labour union k sets nominal wage $W_{k,t}$ to maximise the expected average lifetime utility of its members, $\sum_{s=0}^{\infty} (\beta \xi_w)^s \{ MU_{t+s}^e(W_{k,t+s}/P_{t+s}^e)N_{k,t+s} - MV_{t+s}^eN_{k,t+s} \}$, subject to the labour demand condition $N_{k,t+s} = (W_{k,t+s}/W_{t+s})^{-\theta_w}N_{t+s}$ and the wage indexing rule $W_{k,t+s} = W_{k,t} \times \prod_{u=1}^{s} [\Pi^{\iota_w}(\Pi_{t+u-1}^e)^{1-\iota_w}]$.

Using the two constraints to eliminate $N_{k,t+s}$ and $W_{k,t+s}$ in the union's objective yields

$$\sum_{s=0}^{\infty} (\beta \xi_w)^s \left[\begin{array}{c} \left(P_{t+s}^e \right)^{-1} (W_{k,t})^{1-\theta_w} \left(\prod_{u=1}^s \Pi^{\iota_w} (\Pi_{t+u-1}^e)^{1-\iota_w} \right)^{1-\theta_w} N_{t+s}^e (W_{t+s}^e)^{\theta_w} M U_{t+s}^e \\ - (W_{k,t})^{-\theta_w} \left(\prod_{u=1}^s \Pi^{\iota_w} (\Pi_{t+u-1}^e)^{1-\iota_w} \right)^{-\theta_w} N_{t+s}^e (W_{t+s}^e)^{\theta_w} M V_{t+s}^e \end{array} \right].$$

Let W_t^* denote the nominal wage chosen by all unions that can reset their wages in period *t*. The first-order condition for W_t^* can be written as

$$W_{t}^{*} = \frac{\theta_{w}}{\theta_{w} - 1} \frac{\sum_{s=0}^{\infty} (\beta \xi_{w})^{s} \left(\prod_{u=1}^{s} \Pi^{\iota_{w}} (\Pi_{t+u-1}^{e})^{1-\iota_{w}}\right)^{-\theta_{w}} N_{t+s}^{e} (W_{t+s}^{e})^{\theta_{w}} M V_{t+s}^{e}}{\sum_{s=0}^{\infty} (\beta \xi_{w})^{s} (P_{t+s}^{e})^{-1} \left(\prod_{u=1}^{s} \Pi^{\iota_{w}} (\Pi_{t+u-1}^{e})^{1-\iota_{w}}\right)^{1-\theta_{w}} N_{t+s}^{e} (W_{t+s}^{e})^{\theta_{w}} M U_{t+s}^{e}},$$

which in turn can be rearranged to

$$w_{t}^{*} = \frac{\theta_{w}}{\theta_{w} - 1} \frac{\sum_{s=0}^{\infty} (\beta \xi_{w})^{s} \left(\prod_{u=1}^{s} \frac{\Pi^{t_{w}} (\Pi_{t+u-1}^{e})^{1-t_{w}}}{\Pi_{t+u}^{e}} \right)^{-\theta_{w}} N_{t+s}^{e} (w_{t+s}^{e})^{\theta_{w}} M V_{t+s}^{e}}{\sum_{s=0}^{\infty} (\beta \xi_{w})^{s} \left(\prod_{u=1}^{s} \frac{\Pi^{t_{w}} (\Pi_{t+u-1}^{e})^{1-t_{w}}}{\Pi_{t+u}^{e}} \right)^{1-\theta_{w}} N_{t+s}^{e} (w_{t+s}^{e})^{\theta_{w}} M U_{t+s}^{e}},$$

by using $P_{t+s}^e = P_t \times \prod_{u=1}^s \prod_{t+u}^e, w_{t+s} = W_{t+s}/P_{t+s}$ and $w_t^* = W_t^*/P_t$.

B Appendix: Numerical solution

In this section, we describe our numerical solution approach.

B.1 Extended path method

Despite its shortcomings (see Section 4), the extended path method provides a natural conceptual starting point and a useful benchmark for our local method.⁵⁴ We first describe the method for the (standard) case with rational expectations. Then, we present a more general version that can handle reflective expectations as well.

Rational expectations Suppose that the economy enters period t = 0 at its deterministic steady state, i.e. $\Omega_{-1} = \Omega$ and $S_{-1} = S$. In period t = 0, the economy is then hit by unanticipated shocks that move the exogenous aggregate variables S_0 and therefore the endogenous aggregate ones Ω_0 away from their respective steady state values. If the economy does not experience shocks in later periods t > 0, the assumption of rational expectations implies perfect foresight. Date-0 beliefs about the transition path back to the steady state must therefore coincide with the actual evolution of the economy, i.e. $\{\Omega_{t+s}^e\}_{s=0}^{\infty} = \{\Omega_t\}_{t=0}^{\infty}$. Assuming the economy will have returned to the steady state at date t = T + 1, the computation of sequence $\{\Omega_t\}_{t=0}^T$ boils down to solving a nonlinear equation system, which can e.g. be accomplished by using a numerical solver in combination with a good initial guess.⁵⁵ While the computation of this sequence can be time-consuming even under rational expectations, especially when there is household heterogeneity, it is straightforward to implement conceptually. Note that this step requires tracking the evolution of the infinite-dimensional wealth distribution $\{\Psi_t(b,z)\}_{t=0}^T$ in a tractable way. As is common in the literature, we do so using a discrete approximation of this distribution based on a finite-dimensional histogram (see Reiter, 2009; Young, 2010; Bayer and Luetticke, 2020).

Now, suppose that we are not interested in the economy's response to unanticipated onetime (MIT) shocks in period t = 0, but in running a stochastic model simulation. In this case, the sequence $\{\Omega_t\}_{t=0}^T$ is only needed to determine the beliefs $\{\Omega_{t+s}^e\}_{s=0}^T$ held by agents at date t, which are necessary to pin down the actual values for the current period t = 0, Ω_0 . To compute the actual values for t = 1, i.e. Ω_1 , we proceed as just described for t = 0, except that the initial values now are given by Ω_0 and S_0 , instead of $\Omega_{-1} = \Omega$ and $S_{-1} = S$. New unanticipated shocks then again hit the exogenous states, S_1 . Under certainty equivalence, i.e. ignoring the possibility of future shocks hitting the economy, perfect foresight applies again and the shocks give rise to a new deterministic sequence $\{S_t\}_{t=1}^T$. Since (rational) beliefs adjust to this new information, one needs to compute a new sequence of beliefs $\{\Omega_{t+s}^e\}_{s=0}^T$ for t = 1. This can again be accomplished by using a nonlinear equation solver to calculate the deterministic sequence $\{\Omega_{t+s}\}_{s=0}^T$ for t = 1, which coincides with beliefs

⁵⁴See Christiano et al. (2016) and Erceg et al. (2021) for recent applications of the extended path method.

⁵⁵For the rational-expectations case, one can e.g. use perturbation methods to obtain a good initial guess, both with as well as without household heterogeneity (see e.g. Dobrew et al., 2021).

 $\{\Omega_{t+s}^e\}_{s=0}^T$ in a rational-expectations equilibrium under certainty equivalence. From this sequence, we again only keep the values Ω_1 , which are the actually realised values in period t = 1 that – together with S_1 – serve as initial values for the next period, t = 2, where we will proceed as for t = 1. We repeat these steps for a sufficiently high number of periods \tilde{T} to obtain times series for the model variables of length \tilde{T} . Based on these time series, we can calculate model statistics for selected variables and assess the quantitative model properties for different monetary policy rules. Note that, since we use a nonlinear equation solver for the computations, it is straightforward to impose the ELB constraint for the monetary policy rule. However, such a nonlinearity slows down the numerical computation.

Reflective expectations Under reflective rather than rational expectations, we can proceed as sketched, except that beliefs $\{\Omega_{t+s}^e\}_{s=0}^T$ do not generally coincide with the actual evolution of the respective variables under perfect foresight (and certainty equivalence) anymore. As a result, we cannot directly solve for the entire sequence as before and – similar to Farhi and Werning (2019) and García-Schmidt and Woodford (2019) – use an iterative procedure to determine Ω_t for each time period *t* instead, taking as given initial values Ω_{t-1} and S_{t-1} . Specifically, we apply the following algorithm:

- 1. Let n^* denote the actual level of cognitive ability that we want the economy to exhibit. Choose a small number $\gamma > 0$. The necessary number of iterations with respect to cognitive ability then is n^*/γ . Set the initial time index to t = 0. Set the initial values to $\Omega_{t-1} = \Omega$ and $S_{t-1} = S$.
- 2. Draw period-*t* shocks ε_t . Given S_{t-1} , ε_t and the law of motion for S_{t+s} , calculate the time path $\{S_{t+s}\}_{s=0}^T$. Set the initial level of reflective reasoning to $n = \gamma$ and initialise beliefs $\{\Omega_{t+s}^{e,n}\}_{s=0}^T$.
- 3. Given Ω_{t-1} , $\{S_{t+s}\}_{s=0}^{T}$ and $\{\Omega_{t+s}^{e,n}\}_{s=0}^{T}$, compute the individual policy functions for the households, firms and unions for $s \in \{0, 1, 2, ..., T\}$. While closed-form solutions are available for the reset price and the reset wage, this is only true for individual household consumption and bond holdings in the RANK model. For the HANK model, one has to compute the individual household policies numerically via backward induction, using a global solution method like the endogenous grid method (see Carroll, 2006).⁵⁶
- 4. Go forward in time to compute the sequence $\{\Omega_{t+s}^n\}_{s=0}^T$:
 - a. Set s = 0 and $\Omega_{t-1}^n = \Omega_{t-1}$.
 - b. Compute Ω_{t+s}^n , given Ω_{t+s-1}^n and the policies computed in Step 3.
 - c. Set s = s + 1.
 - d. If $s \le T$ go to Step 4b. If s > T, go to Step 5.

⁵⁶We also use the endogenous grid method when directly solving for a sequence under rational expectations.

- 5. If $n = n^*$, set $\Omega_t = \Omega_t^n$ and go to Step 6. If $n < n^*$, update beliefs, $\left\{\Omega_{t+s}^{e,n+\gamma}\right\}_{s=0}^T = \gamma \times \left\{\Omega_{t+s}^n\right\}_{s=0}^T + (1-\gamma) \times \left\{\Omega_{t+s}^{e,n}\right\}_{s=0}^T$, as well as the level of reasoning, $n = n + \gamma$, and go to Step 3.⁵⁷
- 6. If $t = \tilde{T}$, stop. If $t < \tilde{T}$, update the time index, t = t + 1, and go to Step 2.

This procedure yields the simulated time series $\{\Omega_t\}_{t=0}^{\tilde{T}}$ which we can use to calculate statistics for model variables of interest. Step 4b requires the use of a nonlinear equation solver. Given that – for each $n \in \{\gamma, ..., n^*\}$ – it is only the variables in period *t* and not an entire sequence that is computed in this step, it usually is not particularly time-consuming on its own. Since it will however be performed n^*/γ times for each t + s, $s \in \{0, 1, 2, ..., T\}$, the computational burden is much higher relative to the rational-expectations computations sketched earlier, even if n^*/γ is low. While one could in principle use the iterative algorithm above to approximate the rational-expectations case for sufficiently high n^* , it usually is hence not advisable to do so as it would be much slower – as well as less numerically stable – compared to the procedure described above.⁵⁸

One can imagine various ways to initialise beliefs for Step 2. However, we make the simplifying assumption that beliefs are anchored at the steady state in each period, i.e. $\Omega_{t+s}^{n,e} = \Omega$ for $n = \gamma$, $s \in \{0, 1, 2, \dots, T\}$ and $t \in \{0, 1, 2, \dots, \tilde{T}\}$, which is consistent with our definition of a reflective-expectations equilibrium in the previous section. This assumption has two main implications for our analysis. First, when shocks are only drawn for the first period and set to zero thereafter, i.e. $\{\varepsilon_t\}_{t=0}^{\tilde{T}} = \{\varepsilon_t, 0, \dots, 0\}$ with $\varepsilon_t \neq 0$, where $\varepsilon_t =$ $(\varepsilon_{\beta,t},\varepsilon_{\lambda,t})$, the algorithm outlined above would imply that the sequence $\{\Omega_t\}_{t=0}^{\tilde{T}}$, obtained after applying the entire algorithm above, coincides with the sequence $\{\Omega_{t+s}\}_{s=0}^{T}$ obtained after applying the steps 1 to 5 only once for t = 0, assuming $T = \tilde{T}$. By contrast, if initial beliefs $\{\Omega_{t+s}^{n,e}\}_{s=0}^{T}$, with $n = \gamma$, could change (endogenously) with *t*, this relationship would not generally hold anymore. Although a time-varying anchoring of beliefs could allow for potentially interesting learning dynamics (see e.g. Bianchi-Vimercati, 2022), we prefer to keep the model analysis simple at this stage. Second, anchoring beliefs at the steady state allows us to apply a local approximation for the reflective-expectations model, facilitating fast model simulation. If initial beliefs were to change as time passes, the local approximation would no longer be appropriate anymore because the anchor for the belief formation would (or at least could) move too far away from the steady state.⁵⁹

⁵⁷This updating rule reflects that the continuous belief-updating process $d\Omega_{t+s}^{e,n}/dn = \Omega_{t+s}^n - \Omega_{t+s}^{e,n}$ can be approximated by $(\Omega_{t+s}^{e,n+dn} - \Omega_{t+s}^{e,n})/dn = \Omega_{t+s}^n - \Omega_{t+s}^{e,n}$. If $dn = \gamma$, this equation can in turn be written as $\Omega_{t+s}^{e,n+\gamma} = \gamma \times \Omega_{t+s}^n + (1-\gamma) \times \Omega_{t+s}^{e,n}$. ⁵⁸However, when there are substantial nonlinearities at the aggregate level, we find that the iterative algorithm

⁵⁸However, when there are substantial nonlinearities at the aggregate level, we find that the iterative algorithm can be more stable compared to an approach that directly computes the entire time path for the model variables as sketched above. For instance, we found this to be the case when considering more sophisticated mone-tary policy rules relative to Section 2, such as asymmetric average inflation targeting or temporary price-level targeting (see e.g. Arias et al., 2020; Erceg et al., 2021).

⁵⁹For $\gamma = 1$, one can also use the algorithm to compute an equilibrium where belief formation is based on level-*k* thinking (see Farhi and Werning, 2019). However, we find that the algorithm is not numerically stable in this case and does not converge to the rational-expectations solution for $n^* \to \infty$.

B.2 Local approximation method

Our local model approximation builds on the insights of Boppart et al. (2018). Assume that the random variables in the model follow log-normal AR(1) processes, with persistence parameter ρ_x and shock standard deviation σ_x , for $x \in \{\beta, \lambda\}$. Now suppose that we have computed a transition path for the model variables after a small unanticipated one-time (MIT) demand shock $\varepsilon_{\beta,t} = h > 0$ has hit the economy in the steady state at date t = 0, giving rise to the (shock-specific) transition path $\{\Omega_{t|\beta}\}_{t=0}^{T}$. Furthermore, suppose that $R_{ELB} = 0$ and – for simplicity – only demand shocks are operative in the model, such that we have $\{\hat{S}_{t+s}\}_{s=0}^{T} = \{\hat{\beta}_{t+s}\}_{s=0}^{T} = \{\varepsilon_{\beta,0}, \rho_{\beta}\varepsilon_{\beta,0}, \rho_{\beta}^{2}\varepsilon_{\beta,0}, \rho_{\beta}^{3}\varepsilon_{\beta,0}, \dots, \rho_{\beta}^{T}\varepsilon_{\beta,0}\}$. The hat symbol indicates that a variable is expressed in log-deviations from steady state, $\hat{X}_{t} = \log(X_t/X)$. Following Boppart et al. (2018), we can use

$$\hat{X}_t \cong \sum_{s=0}^T \hat{X}_{s|\beta} \varepsilon_{\beta,t-s},$$

as a linear perfect-foresight approximation around the steady state, for $X \in \Omega$, where $\hat{X}_{s|\beta} = (X_{s|\beta} - X)/h$, for $s \in \{0, 1, 2, ..., T\}$.⁶⁰ Note that if the model is linear and $T = \infty$, the relationship above holds with equality, not just approximately.

To simulate the model, we can now simply draw a sequence of demand shocks and then calculate the model variables' values for each period by using the impulse response coefficients $\{\hat{X}_{t|\beta}\}_{t=0}^{T}$ and the (truncated) shock history for that period via the formula above. Instead of relying on a conventional perturbation of the model based on a recursive formulation, we thus perturb the model in the sequence space, i.e. with respect to the history of past and present aggregate shocks.

There are two key advantages of using this local approximation method in this paper. First, when a model features substantial heterogeneity across agents, the histogram used to approximate the distribution of agents in the economy involves a large number of bins to achieve a reasonable degree of numerical accuracy. As a result, the size of the minimal state space required for a local approximation based on a recursive model formulation is usually going to exceed the number of shocks needed for a model approximation in the (truncated) sequence space in this case. Boppart et al. (2018)'s approach exploits this property and makes it possible to quickly solve and simulate heterogeneous-agent models, requiring only the computation of an impulse response function for each type of aggregate shock, $x \in$ { β, λ }.⁶¹ For the full model with all three types of shocks, one can then approximate the

⁶⁰As Boppart et al. (2018) explain, this normalised transition path represents an impulse response function (IRF) that can be used as a numerical derivative of the time path for variable X with respect to a one-time shock. Loosely speaking, the IRF provides a linearisation of the nonlinear model around the deterministic steady state. If the shock is sufficiently small (and the time horizon T sufficiently long), this linearisation yields the same results as the method by Reiter (2009).

⁶¹Given that the required transition paths need to be calculated only for small unanticipated one-time shocks, the computational cost of this step is not that high, especially under rational expectations.

model variables in period t via

$$\hat{X}_t \cong \sum_{s \in \{0,1,\dots,T\}} \sum_{x \in \{\beta,\lambda\}} \hat{X}_{s|x} \boldsymbol{\varepsilon}_{x,t-s},$$

for $X \in \Omega$.

Second, it is straightforward to apply the sequence-space approximation for models with iterative belief formation (see e.g. Farhi and Werning, 2019; García-Schmidt and Woodford, 2019; Bianchi-Vimercati et al., 2021) because the belief-formation process is directly taken into account when computing the sequences $\{\Omega_{t|x}\}_{t=0}^{T}$.⁶² However, note that, as mentioned earlier, these sequences would be computed under the assumption that beliefs are anchored at the steady state for all periods *t*.

B.3 Enforcing the ELB constraint

The local approximation discussed in the previous section does not take into account the nonlinearity introduced by the occasionally-binding ELB constraint. Building on Bodenstein et al. (2013) and Holden (2016), we enforce this constraint by using anticipated monetary policy shocks.⁶³ In principle, the method is only guaranteed to recover the true solution for a linear dynamic model with an occasionally-binding constraint if expectations are rational and perfect foresight applies. Under these assumptions, enforcing the ELB for the expected evolution of the economy then amounts to enforcing it for the actual evolution of the economy as well. This link is, however, broken for our model with non-rational expectations. Nevertheless, as shown below, we find that one can still apply the method and that the resulting approximation is quite good.

Anticipated monetary policy shocks To apply the method, replace the relationship $R_t = \max \{\tilde{R}_t, R_{ELB}\}$ with $R_t = \tilde{R}_t \prod_{s \in \{0,1,\dots,T\}} \exp(\varepsilon_{R,t-s|t})$, where $\varepsilon_{R,t-s|t}$ is an anticipated monetary policy shock that is announced in period t - s but realised in period t.⁶⁴ Before we can start the model simulation, we need to compute the T + 1 transition paths $\{\Omega_{t+s|R,k}\}_{s=0}^T$, $k \in \{0,\dots,T\}$, using the methods described earlier. The sequence $\{\Omega_{t+s|R,k}\}_{s=0}^T$ denotes the transition path for the economy after the anticipated monetary policy shock $\varepsilon_{R,t|t+k} = h_R > 0$ is revealed in period t ("today") to hit the economy in period t + k. Based on the computed sequences, we calculate the impulse response coefficients $\{\hat{X}_{s|R,k}\}_{s=0}^T$ with $\hat{X}_{s|R,k} = (X_{t+s|R,k} - X) / h_R$, for $k \in \{0, \dots, T\}$ and $X \in \Omega$.

 $^{^{62}}$ By contrast, at least for the HANK model, it does not appear to be feasible to apply standard perturbation techniques to a nonlinear model with iterative belief formation. As in Bianchi-Vimercati et al. (2021), for quite simple models – like the RANK model in this paper – a feasible option would be to first log-linearise the model equations analytically for a given level of reflective reasoning and then apply the iterative belief formation process (see also Bersson et al., 2019; Molavi, 2022).

⁶³Laséen and Svensson (2011) have pioneered the use of such anticipated monetary policy shocks to implement given (exogenous) interest rate paths for linearised DSGE models.

⁶⁴Remember that it is the shadow rate \tilde{R}_t , not the realised rate R_t , which affects \tilde{R}_{t+1} as a lagged argument via the interest rate rule. As a result, the shadow rate \tilde{R}_t is not directly affected by the monetary policy shocks, as they are introduced to replace the ELB and not as an independent source of uncertainty.

Rational expectations We first illustrate the method for the rational-expectations case, building on the ideas of Bodenstein et al. (2013) and Holden (2016) but relying on a local approximation of the model in the sequence space instead. Suppose we want to find the values for the model variables in a given simulation period t. First, use the local method sketched in Section B.1 to generate the unconstrained (log-linear) perfect-foresight sequence $\{\bar{\Omega}_{t+s}\}_{s=0}^{T}$ by using $\bar{X}_{t+s} = \sum_{m \in \{0,1,\dots,T\}} \sum_{x \in \{\beta,\lambda\}} \hat{X}_{m|x} \varepsilon_{x,t+s-m}$, for $X \in \Omega$, initially setting $\varepsilon_{R,t|t+k} = 0$ for all k. Second, check whether the sequence $\{\bar{R}_{t+s}\}_{s=0}^{T}$ violates the ELB constraint, i.e. whether $\bar{R}_{t+s} \ge \log(R_{ELB}/R)$ holds for all future periods $s \in \{0, 1, \dots, T\}$. If this condition is violated for at least one period s, we look for anticipated monetary policy shocks $\{\varepsilon_{R,t|t+k}\}_{k=0}^T$, such that $\hat{R}_{t+s} \ge \log(R_{ELB}/R)$ holds for $\hat{R}_{t+s} = \bar{R}_{t+s} + \sum_{k \in \{0,1,\dots,T\}} \hat{R}_{s|R,k} \varepsilon_{R,t|t+k}$, with $\varepsilon_{R,t|t+k} \ge 0, k \in \{0,1,\dots,T\}$, as well as $[\hat{R}_{t+s} - \hat{R}_{t+s}]$ $\log(R_{ELB}/R)$ $\varepsilon_{R,t|t+s} = 0.65$ We thus need to find non-negative anticipated monetary policy shocks for period t, such that the resulting perfect-foresight transition path for the policy rate, $\{\hat{R}_{t+s}\}_{s=0}^{T}$, (i) does not violate the ELB constraint and (ii) only features strictly positive policy shocks, $\varepsilon_{R,t|t+s} > 0$, in periods $s \in \{0, 1, ..., T\}$ with a binding ELB constraint. Having found appropriate period-t policy shocks $\{\varepsilon_{R,t|t+s}\}_{s=0}^{T}$, we can compute the entire perfect-foresight path consistent with the ELB constraint, $\{\hat{\Omega}_{t+s}\}_{s=0}^{T}$, by using $\hat{X}_{t+s} = \bar{X}_{t+s} + \sum_{k \in \{0,1,\dots,T\}} \hat{X}_{s|R,k} \varepsilon_{R,t|t+k}$, for $X \in \Omega$. Having calculated these sequences, we then only use the first entry, $\hat{\Omega}_t$, as it contains the actually realised values for the model variables in simulation period t. Next, we move to the simulation period t + 1 and proceed as just described for period t.

There are several ways to obtain anticipated policy shocks consistent with the two requirements listed above. Whereas Bodenstein et al. (2013) employ an iterative guessand-verify scheme that shares similarities with Guerrieri and Iacoviello (2015)'s piecewiselinear method, Holden (2016) proposes an algorithm that solves a mixed-integer linear programming (MILP) problem.⁶⁶ Although we find that both methods lead to the same solution for our model, the MILP approach has a number of appealing advantages. Specifically, Holden (2016) proves that a linear dynamic model augmented with anticipated monetary policy shocks yields the same model dynamics as the original model that is linear except for the max-operator. In general, the existence and the uniqueness of a sequence $\{\hat{R}_{t+s}\}_{s=0}^{T}$ consistent with requirements (i)-(ii) are, however, not guaranteed and depend on the local model dynamics, as captured by the impulse response coefficients $\{\hat{R}_{s|R,k}\}_{s=0}^{T}$. Since the approach used by Bodenstein et al. (2013) simply assumes the existence of suitable anticipated monetary policy shocks and then proceeds under this assumption, it is not obvious whether a lack of convergence of their method would be due to numerical issues or the nonexistence of a solution (see Holden, 2016). By contrast, the method proposed by Holden (2016) has the appealing feature of being able to detect whether a solution exists.⁶⁷ In ad-

⁶⁵Formally, finding a shock sequence that satisfies these conditions is a linear complementarity problem (see Holden, 2016).

⁶⁶See Bodenstein et al. (2009) for details on the relationship between the iterative approaches used by Bodenstein et al. (2013) and Guerrieri and Iacoviello (2015).

⁶⁷For our model, we found that the approach presented in Holden and Paetz (2012), which determines the

dition, it makes it possible to select a particular solution in case of multiplicity, e.g. the solution that minimises the length of an ELB spell.

In the remainder, we closely follow Holden (2016) and refer the interested reader to that paper for a detailed treatment of the method. Define the vector $\mathbf{q} = (\bar{R}_t, \bar{R}_{t+1}, \dots, \bar{R}_{t+T})' - (\log(R_{ELB}/R), \log(R_{ELB}/R), \dots, \log(R_{ELB}/R))'$ and the coefficient matrix $\mathbf{M} = (\mathbf{m}_0, \mathbf{m}_1, \dots, \mathbf{m}_T)$, which contains the IRFs $\mathbf{m}_s = (\hat{R}_{s|R,0}, \hat{R}_{s|R,1}, \dots, \hat{R}_{s|R,T})'$. For a given scalar $\boldsymbol{\omega} > 0$, we can find suitable policy shocks $\{\varepsilon_{R,t|t+k}\}_{k=0}^T$ by formulating a MILP problem that solves for scalar $\boldsymbol{\alpha}$, vector $\mathbf{v} = (v_0, v_1, \dots, v_T)'$, and vector $\mathbf{z} \in \{0, 1\}^{T+1}$:

$$\max_{\{\alpha, v, z\}} \alpha \quad \text{s.t.} \quad 0 \le \alpha, \ 0 \le v \le z, \ 0 \le \alpha q + Mv \le \omega \left(\mathbf{1}_{(T+1) \times 1} - z \right).$$

This type of optimisation problem is well understood in the literature and can be solved efficiently using readily available numerical solvers.⁶⁸ How can we recover the anticipated monetary policy shocks from a solution to this problem? If the solution involves $\alpha = 0$, there is no collection of shocks $\{\varepsilon_{R,t|t+s}\}_{s=0}^{T}$ that satisfies the two requirements (i) and (ii) listed above. If $\alpha > 0$, we can recover the shocks via $\varepsilon_{R,t|t+s} = v_s/\alpha$, for $s \in \{0, 1, ..., T\}$. In case different policy shocks are consistent with different model dynamics at the ELB, reflecting equilibrium multiplicity, one can select a particular equilibrium via parameter ω . For instance, $\omega \to 0$ would select anticipated policy shocks that minimise $||q + Mv||_{\infty}$, whereas $\omega \to \infty$ would select the shock sequence minimising $||v||_{\infty}$. In addition, by treating ω as an i.i.d. random variable, it is also possible to use sunspot shocks as an equilibrium selection device for the simulation.

Reflective expectations So far, the method described above relied on perfect foresight to enforce the ELB constraint for the linearised model. However, under reflective expectations, perfect foresight does not hold anymore, such that the equivalence between the realised sequences and the expected sequences breaks down. It is therefore not obvious whether and, if so, how one can use anticipated monetary policy shocks to enforce the ELB in this case. In principle, one would now have to enforce the ELB constraint for all levels of reflective reasoning, $n \in {\gamma, 2\gamma, ..., n^*}$, considered for the belief formation process. Doing so would, however, make the local method as slow as the nonlinear one and therefore undo its key advantage. Instead, we proceed as in the rational-expectations case by assuming that the (unconstrained) actual evolution of the economy populated by agents with cognitive ability n^* , as predicted by the impulse response coefficients and the current history of shocks, coincides with what these agents believe about their economy in general and the interest rate path in particular.

For our purposes, we consider this assumption reasonable for the following reasons. First, if γ is close to zero, the two sequences $\{\Omega_{t+s}^{n^*}\}_{s=0}^T$ and $\{\Omega_{t+s}^{n^*,e}\}_{s=0}^T$ are very close to each other for all $s \in \{0, 1, ..., T\}$. As a result, the sequence $\{\hat{\Omega}_{t+s}^{n^*}\}_{s=0}^T$ provides a

required anticipated policy shocks by solving a quadratic programming problem, also leads to the same results as the other two approaches. However, as in Bodenstein et al. (2013), it assumes the existence of a solution.

⁶⁸We use MATLAB's built-in routine intlinprog.

reasonable approximation of beliefs $\{\hat{\Omega}_{t+s}^{n^*,e}\}_{s=0}^T$. Motivated by this insight, we simply use the approach sketched above for the rational-expectations case also for the reflectiveexpectations case, enforcing the ELB based only on the unconstrained log-linear realised sequence $\{\bar{\Omega}_{t+s}\}_{s=0}^T$. Second, regardless of whether one considers our conceptual argument compelling or not, we find that by enforcing the ELB only based on $\{\bar{\Omega}_{t+s}\}_{s=0}^T$, the resulting sequence $\{\hat{\Omega}_{t+s}\}_{s=0}^T$ consistent with the ELB constraint is quite close to the "ELB-consistent" sequence that a nonlinear perfect-foresight solution would deliver.⁶⁹ This observation justifies our approach from a practical perspective and underscores our conceptual argument.⁷⁰

Capturing nonlinearities at the ELB Whereas Bodenstein et al. (2013) and Holden (2016) – as well as Guerrieri and Iacoviello (2015) – use a recursive model formulation as is common in the DSGE literature, we use the sequence representation described above. Doing so offers an additional degree of flexibility, given by the shock size h_R , that can be used to improve the accuracy of the model approximation for periods at the ELB. Usually, the unconstrained model violates the ELB constraint only if the economy experiences a particularly large shock or a sequence of adverse shocks, likely pushing the economy far away from the deterministic steady state. A binding ELB constraint would then amplify these adverse conditions even further, causing a linear approximation in the neighbourhood of the steady state to possibly deliver a poor performance. As Reiter (2018) highlights, the approximation proposed by Boppart et al. (2018) may perform better if the impulse response coefficients are calculated for h-values that are not close to zero. Whether this is indeed the case depends on how nonlinear the original model is and the properties of the stochastic processes, which affect the time spent close to the steady state. Regarding the anticipated monetary policy shocks, one has good reasons to believe that they are strictly positive only when the economy is far away from the steady state. As a result, it is likely that one can improve the accuracy of the approximation by choosing h_R to keep the difference between the nonlinear solution path and the local one low for selected model variables of interest, such as output or inflation, when the economy is at the ELB.

Accuracy of the local solution To assess the numerical performance of our local model approximation, we compute impulse responses for a large demand shock ($\varepsilon_{\beta,t} = 3\sigma_{\beta}$) with our local method and compare them to impulse responses computed with the nonlinear approach sketched in Section B.1. Figures 8 to 10 show the results for the case without an ELB constraint. Although the size of the shock is quite large, the responses of the interest rate, inflation and output hardly differ across the two solution methods. This observation

⁶⁹We provide details on the numerical accuracy of the local method at the end of this section.

⁷⁰We also assume that the monetary policy shocks, unlike shocks to S_t , are not directly observable by the agents. They are hence only "pseudo-anticipated shocks". This assumption reflects that (i) reflective agents form imperfect expectations about whether the ELB is going to bind or not due to lack of perfect foresight, and (ii) monetary policy shocks do not constitute an exogenous source of aggregate volatility in the model but are determined endogenously to enforce a binding ELB.

holds regardless of assumptions made about cognitive ability and market completeness. With an ELB constraint (see Figures 11 to 13), we notice some small differences for low cognitive ability levels, $n^* \in \{0.5, 1, 2\}$. For low n^* -values, beliefs held by agents and actual outcomes differ to a greater degree. As a result, since Holden (2016)'s approach assumes perfect foresight, enforcing the ELB via anticipated monetary policy shocks works less well in this case than for high n^* -values, reflecting that the model departs more from this assumption. By contrast, when agents are close to being rational, the gap between beliefs and outcomes shrinks and the impulse responses implied by the two solution methods are again very close to each other.⁷¹ Interestingly, for a given n^* , the local solution approach works similarly well in the RANK and HANK models with and without an occasionally-binding ELB. This finding could suggest that our model does not feature sizable nonlinear interactions between the ELB and household heterogeneity.⁷²

⁷¹We use $\gamma = 0.1$ for the computations in this paper (see Section B.1). By lowering γ , the gap between beliefs and outcomes can further be reduced. This in turn lowers the differences between the local and nonlinear impulse responses observed for low *n*^{*}-values under an occasionally-binding ELB. However, using lower γ -values did not affect our qualitative findings and had a negligible effect on our quantitative results, while increasing the computation time.

⁷²We only consider an MIT shock and therefore do not capture the type of interactions studied by Schaab (2020) and Fernández-Villaverde et al. (2021) for HANK models with aggregate risk.



Figure 8: Response of R_t to a contractionary demand shock (without ELB)

Notes: Responses are computed under the IT rule and expressed in percentage deviations from steady state.





Notes: Responses are computed under the IT rule and expressed in percentage deviations from steady state.



Figure 10: Response of Π_t to a contractionary demand shock (without ELB)

Notes: Responses are computed under the IT rule and expressed in percentage deviations from steady state.



Figure 11: Response of R_t to a contractionary demand shock (with ELB)

Notes: Responses are computed under the IT rule and expressed in percentage deviations from steady state.





Notes: Responses are computed under the IT rule and expressed in percentage deviations from steady state.



Figure 13: Response of Π_t to a contractionary demand shock (with ELB)

Notes: Responses are computed under the IT rule and expressed in percentage deviations from steady state.

C Appendix: Additional results

	ELB incidence		Infla	Inflation (%)		put (%)
Policy rule	Freq. (%)	Avg. duration (quarters)	Avg.	Std.	Avg.	Std.
IT	16.29	5.60	1.94	0.64	-0.07	1.10
AIT4	14.31	5.80	1.97	0.59	-0.05	1.06
AIT8	13.99	5.67	1.98	0.59	-0.05	1.06

Table 6: Results for RE-HANK model with real household debt

	ELB incidence		Inflation (%)		Output (%)	
Policy rule	Freq. (%)	Avg. duration (quarters)	Avg.	Std.	Avg.	Std.
IT	20.60	6.92	1.94	0.95	-0.12	1.19
AIT4	17.26	6.67	1.98	0.79	-0.07	1.15
AIT8	16.82	6.42	1.99	0.77	-0.06	1.15

Table 7: Results for RE-RANK model with different price / wage indexation ($\iota_p = \iota_w = 0.9$)

	ELB incidence		Inflation (%)		Output (%)	
Policy rule	Freq. (%)	Avg. duration (quarters)	Avg.	Std.	Avg.	Std.
IT	16.90	5.57	1.92	0.71	-0.08	1.12
AIT4	14.49	5.80	1.96	0.63	-0.05	1.08
AIT8	14.12	5.64	1.97	0.63	-0.05	1.07

Table 8: Results for RE-HANK model with different price / wage indexation ($\iota_p = \iota_w = 0.9$)

	ELB incidence		Infla	Inflation (%)		Output (%)	
Policy rule	Freq. (%)	Avg. duration (quarters)	Avg.	Std.	Avg.	Std.	
IT	17.28	5.58	1.92	0.67	-0.09	1.17	
AIT4	15.08	5.90	1.96	0.60	-0.06	1.10	
AIT8	14.72	5.75	1.97	0.60	-0.05	1.10	

Table 9: Results for RE-HANK model with countercyclical income risk ($\zeta = -1$)

	ELB incidence		Inflation (%)		Output (%)	
Policy rule	Freq. (%)	Avg. duration (quarters)	Avg.	Std.	Avg.	Std.
IT	15.90	5.31	1.95	0.63	-0.07	1.06
AIT4	13.85	5.72	1.97	0.58	-0.05	1.03
AIT8	13.52	5.59	1.98	0.58	-0.04	1.02

Table 10: Results for RE-HANK model with procyclical income risk ($\zeta = 1$)



Figure 14: ELB episode under reflective expectations ($n^* = 1$)

Notes: IRFs are for a large demand shock ($\varepsilon_{\beta,t} = 5\sigma_{\beta}$) and in percentage deviations from steady state.



Figure 15: ELB episode under reflective expectations $(n^* = 2)$

Notes: IRFs are for a large demand shock ($\varepsilon_{\beta,t} = 5\sigma_{\beta}$) and in percentage deviations from steady state.



Figure 16: ELB episode under reflective expectations ($n^* = 3$)

Notes: IRFs are for a large demand shock ($\varepsilon_{\beta,t} = 5\sigma_{\beta}$) and in percentage deviations from steady state.



8

7

6

5

BR HHs







(b) ELB Duration (in Quarters)

BR Firms BR Unions



Notes: The non-rational type of agent is assumed to possess cognitive ability n = 2.





8

7

6

5

BR HHs





(b) ELB Duration (in Quarters)

BR Firms BR Unions



Notes: The non-rational type of agent is assumed to possess cognitive ability n = 2.



Figure 19: Results for BR-RANK without interest rate smoothing ($\rho_R = 0$)





Figure 20: Results for BR-HANK without interest rate smoothing ($\rho_R = 0$)







